MODELLING SUCCESSION PLANNING FOR FEMALE LEADERSHIP IN SAUDI ARABIA HIGHER EDUCATION INSTITUTIONS

DINA ABDULLAH DAHLAN

UNIVERSITI TEKNOLOGI MALAYSIA

MODELLING SUCCESSION PLANNING FOR FEMALE LEADERSHIP IN SAUDI ARABIA HIGHER EDUCATION INSTITUTIONS

DINA ABDULLAH DAHLAN

A thesis submitted in fulfilment of the requirements for the award of the degree of Doctor of Philosophy

Azman Hashim International Business School Universiti Teknologi Malaysia

DEDICATION

This work is dedicated to my beloved father

Dr. Abdulah Sadig Dahlan

ACKNOWLEDGEMENT

Undertaking this Ph.D. has been a truly life-changing experience for me. However, it would not have been possible without the support and guidance I received from many people who contributed and extended their valuable assistance in preparing and completing this study.

First and foremost, my extreme gratitude to my supervisors for their excellent guidance, caring, patience, sincerity, and encouragement that I am indebted to and will never forget. Prof Rosmini Omar's invaluable advice, continuous support, encouragement, and patience during my Ph.D. study. Prof Suzilawati Kamarudin and her invaluable supervision, support, and tutelage during my Ph.D. journey. This Ph.D. would not have been professionally achievable without their guidance and constant feedback. Finally, a special thank you to those who participated in the study during data collection and made an essential contribution towards my Ph.D.; without them, there would not be a study.

A deep heartfelt thank you to my great mentor, role model, and inspiration, Dr. Abdullah Sadig Dahlan. The greatest father there is. He has always provided me with unconditional love and care. He is my greatest supporter and motivator. In addition, I give my deepest love and appreciation to my mother, Sahar Mohamad Mously. She has continuously supported me throughout my life, motivating, cheering, and encouraging me, especially throughout my Ph.D. journey. I would not have made it this far without my beloved parents believing in me and encouraging me to follow my dreams. My accomplishments and success are because they believed in me. Without them and their prayers for me, I would not have had the courage to embark on this arduous Ph.D. journey.

A huge thank you to my adorable angels, Youssif and Hussain, my sons, who have been loving, patient, and supportive, making it possible for me to complete my Ph.D. journey. These past several years have not been easy, both academically and personally, but I persevered with my family and friends' love, motivation, and support. My very profound gratitude to them for providing me with unfailing support and continuous encouragement throughout my years of study.

I am extremely grateful to my cousin Dr. Samar Shoaib for her support, motivation, guidance, and assistance every step of the way of my Ph.D. Her sincere suggestions and assistance during the planning and development of my study were extremely valuable and beneficial. She helped me persevere and remain relaxed, calm, and focused during the hard times of my thesis. Her willingness to give her time and expertise so generously is very much appreciated. I am fortunate to have come this far, but I could not have done it without my family and their tremendous understanding and encouragement. Last but not the least, above all of us, Allah, for answering my prayers and giving me the strength to stride on. Thank you.

ABSTRACT

Despite the significant contributions of females to uplift the economy and well-being of society, they face challenges to being equal compared to their male counterparts. Females still lack support to secure leadership positions, especially in higher education institutions. The concept of succession planning is less likely to be considered an important component of strategizing the organizational plan, particularly in the case of the education sector, with the inclusion of females in some contexts. Thus, it is important to understand the current situation of female leadership aligned with the role of succession planning in the advancement of females in leadership positions in higher education institutions. Accordingly, the purpose of this study is to explore the phenomena and processes of how females could inclusively join leadership positions and succession planning in higher education institutions. Conducting interviews with 23 academicians and experts from Saudi Arabia's higher education institutions, qualitative data analysis is performed using NVivo 12. The findings revealed that females are underrepresented in leadership roles due to the several challenges that hinder their career growth, i.e., national, organizational and individual level barriers. Additionally, succession planning which is a useful tool for closing the gender gap, is not being practiced in Saudi Arabia's higher education institutions. Obviously, in the absence of an effective succession plan, the preparation and selection of future leaders remain doubtful and questionable. It is not surprising that a lack of succession planning in a male-dominated society favors males and marginalizes females. In addition to identifying the elements of succession planning, it is also found that there are obstacles to developing and implementing succession planning. High turnover rate, reluctance, resistance, work-life balance, and inactive HR department are among the challenges to practicing succession planning in Saudi Arabia's higher education institutions. This study contributes to the leadership and succession planning knowledge domain as well as the context of higher education institutions and selected cultures. The inclusion of competent females into a succession planning process may overcome various leadership crises and provide leadership opportunities to females. The findings of this study may help higher education institutions develop succession planning strategies with a clear career path and focus on lessening worklife conflict among female employees.

ABSTRAK

Walaupun terdapat sumbangan besar wanita untuk meningkatkan ekonomi dan kesejahteraan masyarakat, mereka menghadapi cabaran untuk setanding dengan lelaki. Wanita masih kurang sokongan untuk mendapatkan jawatan kepimpinan, terutamanya dalam institusi pendidikan tinggi. Konsep perancangan penggantian kurang diambil kira sebagai komponen penting untuk menyusun strategi pelan organisasi terutamanya dalam sektor pendidikan dengan kemasukan wanita dalam beberapa konteks. Oleh itu, adalah penting untuk memahami keadaan semasa kepimpinan wanita sejajar dengan peranan perancangan penggantian dalam memajukan wanita dalam jawatan kepimpinan di institusi pengajian tinggi. Oleh itu, tujuan kajian ini adalah untuk mengkaji fenomena dan proses bagaimana wanita boleh menyertai jawatan kepimpinan dan perancangan penggantian di institusi pengajian tinggi secara inklusif. Temu bual telah dikendalikan dengan 23 orang ahli akademik dan pakar dari institusi pendidikan tinggi Arab Saudi, dan analisis data kualitatif dilakukan menggunakan NVivo 12. Penemuan menunjukkan bahawa wanita kurang diwakili dalam peranan kepimpinan kerana beberapa cabaran yang menghalang pertumbuhan kerjaya mereka iaitu halangan di peringkat nasional, organisasi dan individu. Selain itu, perancangan penggantian yang merupakan alat yang berguna untuk menutup jurang jantina tidak diamalkan di institusi pengajian tinggi Arab Saudi. Jelas sekali, jika tiada pelan penggantian yang berkesan, penyediaan dan pemilihan pemimpin masa depan tetap diragui dan dipersoalkan. Ia tidak menghairankan bahawa kekurangan perancangan penggantian dalam masyarakat dominan lelaki memihak kepada lelaki dan meminggirkan wanita. Selain mengenal pasti unsur-unsur perancangan penggantian, ia juga mendapati bahawa terdapat halangan untuk membangun dan melaksanakan perancangan penggantian. Kadar pusing ganti yang tinggi, keengganan, rintangan, keseimbangan kehidupan bekerja dan jabatan HR yang tidak aktif adalah antara cabaran untuk mengamalkan perancangan penggantian di institusi pengajian tinggi Arab Saudi. Kajian ini menyumbang kepada bidang pengetahuan perancangan kepimpinan dan penggantian serta konteks institusi pengajian tinggi dan budaya terpilih. Kemasukan wanita yang kompeten ke dalam proses perancangan penggantian boleh mengatasi pelbagai krisis kepimpinan dan memberi peluang kepimpinan kepada wanita. Penemuan kajian ini dapat membantu institusi pengajian tinggi untuk mengembangkan strategi perancangan penggantian dengan laluan kerjaya yang jelas dan dengan fokus untuk mengurangkan konflik kerja-kehidupan di kalangan pekerja wanita.

TABLE OF CONTENTS

	TITLE	PAGE
DEC	CLARATION	iii
DEI	DICATION	iv
ACF	KNOWLEDGEMENT	v
ABS	TRACT	vi
ABS	TRAK	vii
TAE	BLE OF CONTENTS	viii
LIST	Γ OF TABLES	xiii
LIST	Γ OF FIGURES	xiv
LIST	Γ OF ABBREVIATIONS	xvi
LIST	Γ OF APPENDICES	xvii
CHAPTER 1	INTRODUCTION	1
1.1	Introduction	1
1.2	Background of the study	1
	1.2.1 Contextual background	3
1.3	Problem Statement	5
1.4	Research Questions	9
1.5	Research Objectives	9
1.6	Scope of the Study	10
1.7	Significance of the study	10
1.8	Operational definition	13
	1.8.1 Succession planning	13
	1.8.2 Leadership	13
	1.8.3 Female Leadership	13
1.9	Organization of the research	14
1.10	Summary	15

CHAPTER 2	LITERATURE REVIEW	17
2.1	Introduction	17
2.2	Systematic Approach to Literature Review	17
2.3	Research Mapping	19
2.4	Theoretical Underpinning of the Study	24
	2.4.1 Feminist Theory	24
2.5	Leadership	26
2.6	Female leadership	29
	2.6.1 Female leadership in Saudi Arabia's higher education	30
	2.6.2 Barriers to female leadership	33
	2.6.3 Barriers to female leadership in Saudi Arabia	35
2.7	Shared leadership	36
2.8	Succession Planning	38
	2.8.1 Succession Planning in Higher Education	41
	2.8.2 Succession Planning in Saudi Arabia's higher education	46
	2.8.3 Elements and process of Succession planning	48
2.9	Summary	52
CHAPTER 3	RESEARCH METHODOLOGY	53
3.1	Introduction	53
3.2	Research Philosophy	53
3.3	Research methodology	58
	3.3.1 Grounded theory	58
	3.3.1.1 Comparison between Glaser and Strauss Grounded Theory	64
	3.3.1.2 The rationale for using grounded theory by Glaser and Strauss (1967)	66
3.4	Research Design and process	67
3.5	Sample size and sampling technique	69
3.6	Interview Participants	71
3.7	Data Collection	72
	3.7.1 Recording interviews	74

		3.7.2	Elicited t	exts	74
		3.7.3	Extant te	xts	75
	3.8	Instru	ments and	Development of Protocol	75
	3.9	Data a	analysis		86
		3.9.1	Coding p	procedure	86
			3.9.1.1	Open coding	86
			3.9.1.2	Focused coding	87
			3.9.1.3	Theoretical coding	88
		3.9.2	Memoing		89
		3.9.3	Sorting a	nd theoretical coding	89
		3.9.4	Writing-	Up	93
	3.10	Resea	rch Reliab	ility and Validity	93
	3.11	Pilot S	Study		94
	3.12	Ethica	al Consider	rations	95
	3.13	Summ	nary		97
СНАРТЕ	R 4	DATA	A ANALY	SIS	99
	4.1	Introd	uction		99
	4.2	Profile	e of the par	rticipants	99
	4.3	Emerg	ged themes	3	102
	4.4	-	How doe an universi	es leadership is perceived in Saudi ities?	106
		4.4.1	Definitio	n of leadership	106
		4.4.2	Leadersh	ip style	108
	4.5	_		the current situation of female udi Arabian universities?	110
		4.5.1	Characte	ristic of female leaders	112
		4.5.2	The need	for female leaders	114
		4.5.3	Participa	tion of female leaders	116
		4.5.4	Acceptar	ace of female leaders	117
			4.5.4.1	Background and personality	118
			4.5.4.2	Leaders' capability	120
			4.5.4.3	Organizational culture	121

		4.5.4.4	The geographical location of the university	122
	4.5.5	Practices	that support female leadership	123
		4.5.5.1	Organizational support	124
		4.5.5.2	Governmental support	125
		4.5.5.3	Family support	126
	4.5.6	Barriers	to female leadership	127
		4.5.6.1	National level	129
		4.5.6.2	Organizational level	130
		4.5.6.3	Individual level	133
4.6	contri	buting to t	extend does succession planning is the advancement of female leaders in niversities?	134
	4.6.1	Definitio	n of succession planning	136
	4.6.2	Presence	of succession planning	137
	4.6.3	Selection	of future leaders	139
		4.6.3.1	Top management	139
		4.6.3.2	Pre-defined ways	141
	4.6.4	Elements	s of succession planning	142
		4.6.4.1	Organizational support	143
		4.6.4.2	Active HR department	144
		4.6.4.3	Skill inventory	145
		4.6.4.4	Technology	145
		4.6.4.5	Incentive	146
		4.6.4.6	Job description and career path	146
		4.6.4.7	Quota	147
		4.6.4.8	Training	148
	4.6.5	Challeng	es to practice succession planning	150
		4.6.5.1	Passive HR department	151
		4.6.5.2	High Turnover and instability	152
		4.6.5.3	Reluctance	153
		4.6.5.4	Resistance to change	154

	4.6.5.5 Work-life balance	155
4.7	Summary of the findings	156
CHAPTER 5	CONCLUSION AND RECOMMENDATIONS	159
5.1	Introduction	159
5.2	Discussion of findings	159
	5.2.1 Discussion of findings on RO1	159
	5.2.2 Discussion of findings on RO2	161
	5.2.3 Discussion of findings on RO3	167
5.3	Implications of the study	172
	5.3.1 Theoretical implication	172
	5.3.2 Practical implications	174
5.4	Limitations and suggestions for future research	178
5.5	Conclusion	179
REFERENCES		181
LIST OF PUBLI	ICATIONS	229

LIST OF TABLES

TABLE NO. TITLE	PAGE
Table 1.1 Female Leaders in Higher Education, Saudi Arabia	4
Table 2.1 Classification of Studies on Succession Planning in Higher Educat	ion 21
Table 2.2 Definitions of succession planning	39
Table 3.1 Comparison between research philosophies	54
Table 3.2 Components of Grounded Theory (complied by researcher)	61
Table 3.3 Comparison between Glaser and Strauss Grounded Theory	65
Table 3.4 Interview questions for Informants in Universities (Insider)	76
Table 3.5 Interview Protocol for Informants Outside Universities (outsider)	81
Table 4.1 Profile of the participants	99
Table 4.2 The emerged themes and sub-themes	102
Table 4.3 characteristics of female leaders	112
Table 4.4 Summary of the research findings	156

LIST OF FIGURES

FIGURE 1	NO. TITLE	PAGE
Figure 2.1	Systematic review steps (Clark and Creswell, 2015)	18
Figure 2.2	Summary chart of the systematic review of literature	19
Figure 2.3	Criteria for selection of relevant studies	20
Figure 2.4	Number of selected papers on the topic of "Succession Planning in	
	Higher Education"	23
Figure 2.5	University Faculty Organizational Structure in Saudi Arabia	28
Figure 2.6	University Organizational Structure Saudi Arabia	29
Figure 2.7	The Seven-Pointed Star Model for Systematic Succession Planning	
	(Rothwell, 2010)	49
Figure 3.1	Research onion model	54
Figure 3.2	Visual illustration of the Grounded Theory (researcher visualization	n) 61
Figure 3.3	Research flow diagram (Source: Clark and Creswell, 2015)	69
Figure 3.4	Sequence of interviews	73
Figure 3.5	Visual presentation of Data collection and Analysis (Charmaz, 2014)	91
Figure 4.1	Themes and subthemes captured from NVivo	111
Figure 4.2	Subthemes relevant to background and personality	118
Figure 4.3	Subthemes of practices that support female leadership	123
Figure 4.4	Barriers to Female leadership	128
Figure 4.5	Barriers to female leadership	128
Figure 4.6	Themes and subthemes captured from NVivo for succession planning	135
Figure 4.7	Themes and subthemes extracted from NVivo relevant to the selection of future leaders	139
Figure 4.8	Factors that affect the implementation of the succession planning	143
Figure 4.9	Chart for challenges to practice succession planning extracted from NVivo	151
Figure 5.1	Theoretical framework	174

Figure 5.2 Practical framework: strategies to improve female leadership by the implementation of the SP

177

LIST OF ABBREVIATIONS

SA - Saudi Arabia

KSA - Kingdom of Saudi Arabia

SP - Succession planning

FL - Female leadership

LIST OF APPENDICES

APPENDIX	TITLE	PAGE
Appendix A	Instrument and interview protocol	219

CHAPTER 1

INTRODUCTION

1.1 Introduction

This chapter presents research exploring female engagement in leadership positions in line with succession planning of higher education institutions in Saudi Arabia. First, the chapter describes the study background in order to identify the research problem, questions, and objectives. Further, the significance and scope of the study elaborate on the possible contributions and clarify boundary and limitations of the current research. Subsequently, this chapter defines the key concepts used to explore the role of the female as a leader in higher education institutions. Overall, this chapter contributes to understanding the holistic picture of a female's situation in a leadership position in higher education.

1.2 Background of the study

The arguments made by neoliberals have placed emphasis on the importance of women's education. It highlights that education is a human right and is essential for creating a moral and progressive society. Women's education is good for society as a whole, but it also provides a great advantage to the economy (Olson-Strom and Rao, 2020). Women in higher education have shown that giving them a seat at the table can be prosperous at both a community and national level.

Despite the significant contribution of females to uplift the economy, they did not achieve equal status compared to men (World Economic Forum, 2018). Studies indicated that females' career positions as professionals and render services as leaders remain underrepresented (Mifsud, 2019; Murray and Mifsud, 2019). Organizations

are failing females in their career trajectories. There is an ongoing and consistent under-representation of females in educational presidency positions (Shaw and Hernandez-gantes, 2021).

Females faced certain challenges in achieving top positions (i.e., gender discrimination and job segregation based on race and status); hence, they are less likely to be involved in professional development than their male counterparts (Alghofaily, 2019). As evident, Gender Inequality Index (GII) published by United Nations Development Programme highlighted that women are discriminated against in health, education, political representation, labor market, and more. Besides, the reports show the loss in potential human development due to inequality between female and male achievements in these dimensions (UNDP, 2020).

Leadership is one of the most critical factors organizations need to consider to achieve their goals (Cobanoglu, 2020). The qualities of a leader and good leadership have paved the way for change, enhanced employees' commitment and performance, and improved overall organizational performance (Anderson, 2017). The literature suggests that most female leaders are concentrated in areas traditionally associated with "women's issues," and their overall percentage remains low (Barraza Vargas, 2019). The study of leadership in the higher education has been limited in the past. Although gender imbalance in senior positions in higher education is neither acceptable nor sustainable, the improvement rate is slow. the slow pace of change achieved by past attempts at developmental initiatives and goal setting should be augmented by a more positive approach to change (Power, 2020). Most women in the higher education sector remain restricted to middle-ranking posts, rising as high as department heads, deans, controllers of examination, and even registrars but hardly ever becoming vice-chancellors (Gandhi and Sen, 2021). The findings of the study by Shaw and Hernandez-Gantes (2021) suggest that female vice presidents outnumber female presidents.

Historically, providing gender equity both nationally and internationally in terms of accessing and achieving educational opportunities is seen as a necessity for developing sustainable succession plans (Dahlvig and Longman, 2020). However, it

is limited to a sustainable succession plan and building leadership capacity and skills throughout the academic staff. Furthermore, most notably, creating an atmosphere that promotes an environment for the growth of all academic leaders (Power, 2020).

Researchers have highlighted the need for leadership roles to be held by capable individuals and for capable leaders always to be ready to take on a role in the event of resignation, retirement, promotions, death, growth, expansion, and creation of new positions. Having a supply of capable leaders shows an organization's true strength (Desarno *et al.*, 2020). Hence, it is vital for every organization to have an effective and sustainable succession planning process that can withstand any future challenges. The most important elements of a succession plan are growth plans, leadership development, and mentoring programs that allow for positive relationships that cultivate, inspire, and assist faculty in developing the necessary skills to succeed in a leadership role (Phillips, 2021).

In light of the vital role of the women in contributing to the social and economic development and also considering the fact that women face challenges in holding leadership position in general and higher education in particular, this study focus on succession planning as a strategic tool that can help for promoting female leaders in society. The present research is concentrated on Saudi Arabia as it is discussed in the following section.

1.2.1 Contextual background

A vacuum of literature on gender and higher education exists outside America (primarily the United States) and Western Europe. The topic is underdeveloped in the Asian context (Olson-Strom and Rao, 2020). In Arab countries, females are more involved in managing operations related to education, administration, and health care (Alghofaily, 2019). In the context of Saudi Arabia, females have been studied well in gender equality, career advancement, and management (Abalkhail, 2017; Syed *et al.*, 2018), yet, a limited number of studies have covered mixed-gender participation in holding leadership positions. Hence, if

females at work in Saudi Arabia are disadvantaged in their practice, gender discrimination occurs based on socio-cultural and religious-cultural factors (Abalkhail, 2017; Fallatah and Syed, 2017).

In comparison with the western culture, there is a need to explore the barriers to female leadership in Saudi Arabia (Abalkhail, 2017; Al-Asfour *et al.*, 2017; Hodges, 2017). It is vital to conduct some pre-impact evaluation research studies to understand the factors that affect the recruitment and selection of senior academic leaders. Along with that, there is a need to point out the challenges females are facing in their advancement to leadership positions, especially in the higher education sector (Alsubaie and Jones, 2017).

Females tend to be excluded from organizational policymaking, planning, and decision-making (Mustafa and Troudi, 2019). The statistics in Table 1.1 prove that female participation in all top positions is far behind men's engagement. The more significant number reported for the dean position at Princess Nourah Bint Abdulrahman University is because the job was open to females only. Females at the vice president and dean positions are also centric to the departments allocated for the female students at King Abdullah University of Science and Technology, King Faisal's, and King Saud's Universities.

Table 1.1 Female Leaders in Higher Education, Saudi Arabia

Positions	Number of Females	Number of Men	
Director of a university	1	33	
Vice president	12	128	
Dean	61	330	
Deputy dean of faculty	228	542	
Total	302	1033	

Source: Ministry of Education (2017)

Although the current situation of female leadership is not satisfactory, the Kingdom's Vision 2030 and its programs have given women's empowerment great attention. One of the most prominent outputs of the Kingdom's Vision 2030 was the formulation of legislation that supports the empowerment of women's rights,

including wages and the provision of job opportunities in all sectors, supporting women in leadership positions, and promoting women's rights (Shura Council, 2021). In addition, some factors favor the exploration of female succession planning practices in higher education in Saudi Arabia. First, females are placed with a gender-segregation approach in which they are involved in managing operations related to greater involvement of females, such as education, administration, and health care (Alsubaie and Jones, 2017; Alghofaily, 2019). Secondly, wider gender gap as Saudi Arabia is ranked 141 out of 149 countries (World Economic Forum, 2018).

Actions that advance women's leadership may be as diverse as the contexts themselves for each context and level. Regardless of the social context, what is essential and common is identifying and addressing barriers through developing strategies that advance gender equity for everyone's benefit (Dahlvig and Longman, 2020). Considering Saudi Arabia's national agenda for strengthening higher education institutions, it is essential for universities to track the progress under effective leadership. Achieving the desired results is becoming vital for overcoming the leadership crisis and practicing succession planning. A volatile financial environment, the rise of international partnerships, greater accountability pressures, the need for new business models, new technologies, and changing demographics are just some of these challenges, which call for leadership solutions in higher education (Kezar, 2017). Research implications may open new doors for practitioners, education consultants, and policymakers to practice succession planning in higher education with the inclusion of females.

1.3 Problem Statement

Female leadership is still in the infancy stage, and more is needed to be done to ensure that they are on equal footing as their male counterparts. The current research agenda has advocated for more female representation and highlighted the positive impacts of female leaders on the community. Despite this recognition, women still do not have equal access to opportunities (Tran, 2020). The

underrepresentation of female leaders in leadership positions is a global phenomenon, not specific to a particular geographical context. To illustrate, findings of the study by Dahlvig and Longman (2020) revealed that women who aspire to senior- level leadership in the United States continue to find their professional journeys hindered by a variety of internal, organizational, and broader cultural barriers. Similarly, Gandhi and Sen (2020) found that female leadership in the top positions in academia in India is not just under-represented but also under-researched. Similarly, in the Vietnam context, Tran (2020) found evidence for the underrepresentation of women, especially in high-level leadership positions.

Despite the recent rapid growth in higher education worldwide, gender inequality in higher education remains a significant issue (Olson-Strom and Rao, 2020). The number of women in leadership positions, especially high-level ones, is still far from being equal with that of men (Tran, 2020). Although a few steps have been taken to reduce the gap in gender imbalance in senior positions in higher education, more needs to be done as the current system is slow and ineffective (Power, 2020). Moreover, a bias towards male leadership in higher education has amplified the gender imbalance (Tran, 2020). That being said, the world is moving towards recognizing the importance of gender balance in all elements of the organization. Educational institutions, in particular, have begun to invest more in leadership programs that encourage both men and women to participate (Kairys, 2018). In addition to financial investments, they are also moving towards more nonfinancial investments such as building leadership capacity and skills throughout the academic staff, linking resources to a gender-balanced leadership team, seeding leadership posts in areas of specific gender imbalance are all positive moves to achieve a pool of academic leaders (Power, 2020).

One effective way to expand the internal pool of leadership candidates and fill leadership positions is by developing and implementing succession planning programs (Fusarelli *et al.*, 2018). Although succession planning for business has been thoroughly covered in the literature with a sufficient number of initiatives, strategic designs, and programs for corporate organizations (Rosenthal *et al.*, 2018; Black *et al.*, 2019; Ayandibu and Kaseeram, 2020), it is still a new concept in higher

education and hardly addressed in scholarly research (Gilbert, 2017; Keller, 2018). As a result, succession planning is less likely to be considered a critical component in strategizing the organizational plan, specifically in the case of the education sector (King, 2019; Loomes *et al.*, 2019). The limited studies on the matter of succession planning in higher education was mainly focused on western countries, thereby adding to the gaps in both theory and practice in the background (Gilbert, 2017; Keller, 2018; Aylett, 2019).

When females entered the workplace, they faced complex rules, regulations, and a harsh environment. It also creates a barrier for female to perform their routine work and is less likely to have the freedom to participate in the management decisions (Abalkhail, 2017). The ultimate reason behind this is gender segregation, as specified by several researchers (Al-Asfour *et al.*, 2017; Hodges, 2017; Alghofaily, 2019). Gender-segregated work environments might limit opportunities for gender equality, sustain patriarchal structures, and maintain traditional attitudes regarding female advancement in the work place (Alomair, 2015).

Theorizing the concept of leadership, female as a subject has been studied in higher education settings that covers multiple styles and context (Abalkhail, 2017; Mohnot, 2017; Dopson *et al.*, 2018). Thus, it becomes fragmented because there is inconsistency in results to recommend an appropriate leadership style that strategically support female leadership in the academic institutions. Female leaders in higher education face challenge to play their effective role at management positions (Al-Asfour *et al.*, 2017; Hodges, 2017; Alghofaily, 2019). Unfortunately, the current practices of top-down leadership are impractical to the current landscape in higher education.

Prior literature suggest that current leadership approaches are ineffective to deal with change management and innovative systems (Keller, 2018; Alghofaily, 2019). Eventually, leadership models have pointed out that leaders cannot use one kind of behavior or style to deal with multiple situations. There is a need to adapt different kind of leadership styles to be productive in higher education (Mustafa and Troudi, 2019). In the current era, there is utmost need to guide higher education

institutions with a modern strategic leadership approach that bring effective results in order to achieve the organizational goals, thus old leadership approaches seem ineffective (Washington, 2016; Ritchie, 2020). The drastic change in the higher education environment urges the need to practice shared leadership approach to have collaborative efforts to outcomes (Lu *et al.*, 2017). Today's complex environments require new forms of collaborative or shared leadership to help campuses become nimbler and responsive to needed changes. Higher education leaders now face a very different set of challenges that necessitate new forms of leadership. Some argue that shared leadership not only meets today's challenges, but also is a better fit for higher education (Kezar, 2017).

The new forms of higher education, including massive open online courses, distance learning and artificial intelligence have been changing higher education and create both challenges and opportunities. Beside those global factors, higher education institutions also face their regional and local challenges (Lu *et al.*, 2017). Concerning leadership crisis in higher education, models, framework, and programs related to female succession planning in Saudi Arabia's higher education are limited from a practical approach to a leadership crisis in higher education. Therefore, there is utmost need to fill the literature gap by exploring the succession planning and strategic ways for female engagement to render services at leadership positions in higher education institutions in Saudi Arabia. In the case of Saudi Arabia, researchers have conducted qualitative and quantitative studies on female leadership, females in education, challenges for Saudi females at work and females in the management of higher education (Abalkhail, 2017; Al-Asfour *et al.*, 2017; Hodges, 2017; Alghofaily, 2019; Varshney, 2019).

This current study's systematic review (see Chapter 2) describes the hindrance factors for females from performing their roles in leadership positions, particularly in Saudi Arabia. The following sections provide the study purpose, research questions, and objectives with reference to the research gaps.

1.4 Research Questions

Considering the problems addressed in the previous sections related to females as part of succession planning and engagement in leadership positions in the context of Saudi Arabia, this study explores answers to the following questions:

RQ1: How does leadership is perceived in Saudi Arabian universities?

RQ2: How is the current situation of female leadership in Saudi Arabian universities?

RQ3: To what extend does succession planning is contributing to the advancement of female leaders in Saudi Arabian universities?

1.5 Research Objectives

Based on the research questions, the study sets the following research objectives:

RO1: To discover how leadership is perceived in Saudi Arabian universities

RO2: To understand the current situation of female leadership in Saudi Arabian universities

RO3: To explore whether or not succession planning is contributing to the advancement of female leaders in Saudi Arabian universities

1.6 Scope of the Study

The focus of this study is to explore the current leadership situation in Saudi Arabia's higher education in line with practicing the succession planning, which provides leadership opportunities for females. Mainly, this study attempts to understand the acceptance of public female leaders and their current participation, supporting factors, barriers, and the role of succession planning to ensure equal opportunity for men and women in career growth in higher education.

The study collected data from selective higher education institutions in Saudi Arabia, namely University of Business and Technology, King Abdulaziz University, Sulaiman Al Rajhi university, University of Jeddah, University of Bisha, Umm Al-Qura University, Prince Mohammad Bin Fahd University, Prince Sultan University, Dar Al-Hekma University, King Saud University, and Taif University. In addition to universities, the Ministry of Education and Jeddah chamber of commerce are included in the sampling frame. Participants of this study are deans, deputy/vice deans or heads of departments/schools, instructors (faculty staff), and administrators (line management). In addition, the study selected one education consultant from the Ministry of Education and one representative from the Jeddah Chamber of Commerce.

1.7 Significance of the study

Given this current era of significant change in higher education and a growing concern that existing approaches to leadership are ineffective, there is growing attention to the importance of understanding the leadership required to guide campuses successfully (Kezar, 2017). Consequently, the purpose of this study is to explore the phenomena and process of inclusiveness and engagement of females in leadership positions in line with succession planning within institutions of higher education in Saudi Arabia. More precisely, this study investigates the female leadership crisis using a unique way to explore factors that support and impede females' advancement to leadership positions in Saudi Arabian public and private

universities. Additionally, the role of succession planning in improving female leaders is explored in this study. Researchers have identified that, even when a woman candidate meets the educational requirements, women are still underrepresented at the top of institutional leadership hierarchies. The ecosystem that women function in now has been set up to hold women back and advance men (Diehl and Dzubinski, 2016).

Hopefully, the information and findings of this study could provide a direction for senior leaders to strategize succession planning and deal with leadership crises by giving equal opportunities to female leaders in higher education. This approach supports management personnel in higher education to deploy modern tools and practices for succession planning in higher education. In addition, this study would bridge the gap in the literature, which is still lacking in female succession planning in higher education. Earlier works by researchers focus on leadership styles in academic institutions (Al-bakr et al., 2017; Alghofaily, 2019), workforce management in higher education (Cooper, 2019; Loomes et al., 2019; Wilkins and Neri, 2019), women's career advancement (Hodges, 2017), leadership challenges (Alghofaily, 2019; Mustafa and Troudi, 2019) and gender equality (Syed et al., 2018). However, these studies lack a theoretical framework on higher education succession planning by targeting females in academic settings. Therefore, this study provides a reference point for researchers in various fields concerning leadership crisis, succession planning, and female leadership engagement in higher education.

Furthermore, this study contributes to improving practices in succession planning that focus on diversity and inclusiveness, particularly for females. This study brings the attention of multiple stakeholders to several hidden factors that indicate the female leadership crisis in higher education and help them design preemptive actions to deal with it. More importantly, this study supports management to utilize untapped talent for future leadership positions in higher education. This study's findings also benefit women by raising their awareness of all females' challenges. According to Míltersteíner et al. (2020), when women perceive

that others are experiencing the same difficulties and challenges, they may feel more engaged and aware of typical particularities and possibilities.

As stated earlier, certain studies in literature mention the leadership crisis, female leaders, and career advancement in higher education. Nonetheless, the female succession planning framework that describes the process of developing in-house talent to fulfill the need for the future workforce at management and leadership levels in higher education remains eclipsed in the academic literature. Saunders *et al.* (2015) mentioned that research implications might effectively apply when findings come from a contextual setting. Thus, this study elucidates the practices for succession planning and steps to resolve the female leadership crisis in higher education by Saudi Arabian universities.

The present research is also significant from the methodological perspective. The qualitative method is suitable for exploring and understanding emerging concepts such as succession planning. Using a qualitative method provided an opportunity to investigate the current leadership style, female leadership situation, and succession planning statement in Saudi Arabian context. This is particularly important because there is a need for empirically-based studies on academic leadership in Saudi Arabia's higher education (Alomair, 2015) to ensure the sustainability of effective leadership.

From a contextual perspective, the study findings may encourage higher education institutions in Arab and Gulf countries to practice female succession planning frameworks with respect to their socio-cultural settings. In addition, Saudi Arabia's Vision 2030 aims for education that contributes to economic growth and "aims to have at least five Saudi universities among the top 200 universities in international rankings" (Saudi Vision, 2017). Since Saudi Arabia's government is focusing on gender issues, it is important to provide equal opportunities for both genders (Saudi Vision, 2017; Ministry of Education, 2017; Ministry of Education Statistics Center, 2017). There should be a space for female leadership in the succession planning for Saudi Arabia universities to achieve the desired results. Therefore, the present study's findings may provide a guideline for the government

and practitioners in improving their understanding of the current female leadership situation towards implementing inclusive and equitable succession planning in Saudi Arabia's higher education institutions.

1.8 Operational definition

1.8.1 Succession planning

Succession planning is a strategic effort to ensure skillful, productive, and high-performing employees are retained within the organization for leadership or managerial positions over time (Owolabi and Adeosun, 2021).

1.8.2 Leadership

Leadership is a process that involves people with common goals, and leaders are people who can influence a group to commit willingly to a common goal (Turner, 2018).

1.8.3 Female Leadership

Females in the sphere of higher education provide teaching, administrative, and management support to deploy the best practices to lead the organization in equal standing with men. This refers to a practical approach to females' contribution to leadership positions in higher education institutions (Loomes *et al.*, 2019).

1.9 Organization of the research

This research study proposal comprises five chapters:

Chapter 1 provides the background of the study and describes the contextual information related to females' engagement in higher education and their role in leadership positions. It covers the knowledge gap and problem statement where females face obstacles to securing their leadership positions according to academic responsibilities. It refers to the research questions and objectives elaborated in line with the grey areas identified for this study. This chapter also covers the significance and scope of this study.

Chapter 2 presents the systematic review, and theoretical underpinning of the study domain described as females in higher education leadership. It covers the theoretical argument from published studies and supports the study framework. Further, a systematic review depicts the existence of current knowledge and future research areas to work on.

Chapter 3 presents the study methods and procedures adopted for data collection and analysis. The procedure for data management, a pilot test of instruments, and an analytical approach are included in this chapter. It confirms the validity, authenticity, and reliability of data and supports the generalizability of study findings. Overall, it covers the basic steps that are performed to execute the research study.

Chapter 4 focuses on data analysis procedures and techniques used to analyze the obtained data through interviews. Conducted interviews are transcribed, and emerging themes are discussed in this chapter. Finally, the main findings of this study are presented in this chapter.

Chapter 5 is mainly concerned with discussing research findings and conclusions drawn from the results. The chapter also elaborates on the contributions

of this study to the theory and practice. Finally, the chapter ends with a discussion on limitations and recommendations for future studies.

1.10 Summary

The study rationale is elaborated in this chapter which comprises the background of the study, problem statement, research questions, and objectives. This chapter describes the purpose and significance of the study, which is vital to cover the knowledge gap that exists in the Saudi Arabian higher education institutions. Finally, it provides recommendations for scholars to conduct future studies.

REFERENCES

- Abalkhail, J. M. (2017) 'Women and leadership: challenges and opportunities in Saudi higher education', *Career Development International*, 22(2), pp. 165–183.
- Abalkhail, J. M. and Allan, B. (2015) 'Women's career advancement: Mentoring and networking in Saudi Arabia and the UK', *Human Resource Development International*, 18(2), pp. 153–168.
- Abdullah, Z., Samah, S. A. A., Jusoff, K. and Isa, P. M. (2009) 'Succession Planning in Malaysian Institution of Higher Education', *International Education Studies*, 2(1), pp. 2005–2008.
- Acker, S. and Wagner, A. (2019) 'Feminist scholars working around the neoliberal university', *Gender and Education*, 31(1), pp. 62–81.
- Adewale, O., Abolaji, A. and Kolade, O. (2011) 'Succession planning and organizational survival: Empirical study on Nigerian private tertiary institutions', *Serbian Journal of Management*, 6(2), pp. 231–246.
- Akram, F., Murugiah, L. a/p and Shahzad, A. (2017) 'Cultural Aspects and Leadership Effectiveness of Women Leaders: A Theoretical prospective of Saudi Arabia', *Pakistan Journal of Humanities and Social Sciences*, 5(1), pp. 25–35.
- Al-Asfour, A., Tlaiss, H. A., Khan, S. A. and Rajasekar, J. (2017) 'Saudi women's work challenges and barriers to career advancement', *Career Development International*, 22(2), pp. 184–199.
- Al-bakr, F., Bruce, E., Davidson, P., Schlaffer, E. and Kropiunigg, U. (2017) 'Empowered but not Equal: Challenging the Traditional Gender Roles as Seen by University Students in Saudi Arabia', *Forum For International Research in Education*, 4(1), pp. 52–66.
- Al-jaradat, M. K. M. (2014) 'Challenges Facing Women Academic Leadership in Secondary Schools of Irbid Educational Area', 7(5).
- Al-Ohali, M. and Al-Mehrej, H. (2012) 'Faculty salary and remuneration in the Kingdom of Saudi Arabia.', in *Paying the professoriate: A global comparison of compensation and contracts*, pp. 278–287.

- Al-Sudairy, H. (2017) Modern woman in the Kingdom of Saudi Arabia: Rights, challenges and achievements, Cambridge Scholars Publishing. Cambridge Scholars Publishing.
- Albakry, A. H. (2016) Voices of women leaders in corporate institutions in Saudi Arabia.
- Alexander, N. H. (2011) 'Teaching Leadership to Female Students in Saudi Arabia', *Advancing Women in Leadership*, 31, pp. 199–212.
- Alexander, N. H. (2013) 'Saudi Arabia Female Students' Perception of Effective Female Leaders.', *Advancing Women in Leadership*, 33, pp. 142–150.
- Alghofaily, L. (2019) Investigating the Perceived Barriers to Women's Rise to Leadership in Higher Education in Saudi Arabia.
- Al Alhareth, Yahya, Al Alhareth, Yasra and Al Dighrir, I. (2015) 'Review of Women and Society in Saudi Arabia', *American Journal of Educational Research*, 3(2), pp. 121–125.
- Ali, Z. and Mehreen, A. (2019) 'Understanding succession planning as a combating strategy for turnover intentions', *Journal of Advances in Management Research*, 16(2), pp. 216–233.
- Alimo-metcalfe, B. and Alimo-metcalfe, B. (2010) 'An investigation of female and male constructs of leadership and empowerment'.
- Alomair, M. O. (2015) 'Female Leadership Capacity and Effectiveness: A Critical Analysis of the Literature on Higher Education in Saudi Arabia', *International Journal of Higher Education*, 4(4), pp. 81–93.
- Alotaibi, F., Cutting, R. and Morgan, J. (2017) 'A Critical Analysis of the Literature in Women's Leadership in Saudi Arabia', *International Journal of Business Administration and Management Research*, 3(1), p. 29.
- Alsubaie, A. and Jones, K. (2017) 'An Overview of the Current State of Women's Leadership in Higher Education in Saudi Arabia and a Proposal for Future Research Directions', *Administrative Sciences*, 7(4), p. 36.
- Anderson, E. K. and Solomon, C. R. (2015) Family-friendly policies and practices in academe 2015,.
- Anderson, M. (2017) 'Transformational Leadership in Education: A Review of Existing Literature', *International Social Science Review*, 93(1), p. 4.

- Ayalew, S., Manian, S. and Sheth, K. (2021) 'Discrimination from below: Experimental evidence from Ethiopia', *Journal of Development Economics*, 151, pp. 1–37.
- Ayandibu, A. O. and Kaseeram, I. (2020) 'The Future of Workforce Planning', in *igi-global.com*, pp. 127–156.
- Aylett, A. (2019) 'Influencers of Succession Planning Among Rural Community College Chief Executives', *ProQuest Dissertations and Theses*, p. 97.
- Barden, D. M. (2010) 'Where will you find your future leaders', *Chronicle of Higher Education*, 56(17).
- Barginere, C., Franco, S. and Wallace, L. (2013) 'Succession planning in an academic medical center nursing service', *Nursing Administration Quarterly*, 37(1), pp. 67–71.
- Barnett, R. and Davis, S. (2008) 'Creating Greater Success in Succession Planning', *Advances in Developing Human Resources*, 10(5), pp. 721–739.
- Barraza Vargas, C. (2019) 'Women in Command: The Impact of Female Leadership on International Organisations', *Global Society*. Taylor & Francis, 33(4), pp. 541–564.
- Barton, A. (2019) 'Preparing for Leadership Turnover in Christian Higher Education:

 Best Practices in Succession Planning', *Christian Higher Education*.

 Routledge, 18(1–2), pp. 37–53.
- Barton, T. R. (2006) 'Feminist leadership: Building nurturing academic communities.', *Advancing Women in Leadership Journal*, 20.
- Beheshtifar, M. and Moghadam, M. N. (2011) 'Studying the competency-related models in succession planning', *European Journal of Economics, Finance and Administrative Sciences*, 34(34), pp. 113–121.
- Belknap, R. A. and Cruz, N. (2007) 'When I Was in My Home I Suffered a Lot: Mexican Women' s Descriptions of Abuse in Family of Origin', *Health care for women international*, 28(5), pp. 506–522.
- Bello, M. A. and Thompson, E. S. (2003) 'Recruiting the cataloguers for tomorrow: the need for succession planning in Ghanaian academic libraries', *Library Management*, 24(6), pp. 310–315.
- Berchelman, D. K. (2005) 'Succession planning', *The Journal for Quality and Participation*, 28(3), p. 11.

- Betts, Kristen and Betts, Keith (2009) 'Visible Administrative Career Paths ,
 Professional Development , Succession Planning & Commitment to
 Diversity', 7(2).
- Birks, M. and Mills, J. (2011) *Grounded theory: A practical guide*. Sage publications.
- Bisbee, D. C. and Miller, M. T. (2006) 'Executive succession practices in land grant universities', *Academic Leadership*, 4(3), p. 1.
- Black, S., Gardner, D., Pierce, J. and Steers, R. (2019) 'Talent Development and Succession Planning', *Human Resource Management*.
- Bleich, M. R. (2019) 'The professional development educator and leadership succession planning', *Journal of Continuing Education in Nursing*. Slack Incorporated, 50(3), pp. 104–106.
- Boerner, H. (2015) 'Game plan', Community College Journal, 85(6), p. 20.
- Boggs, G. R. and Kent, E. L. (2002) 'President Academy: An Evolution of Leadership Development', in *New directions for community colleges*, pp. 51–58.
- Bolden, R. and O'Regan, N. (2016) 'Digital Disruption and the Future of Leadership:

 An Interview With Rick Haythornthwaite, Chairman of Centrica and MasterCard', *Journal of Management Inquiry*, 25(4), pp. 438–446.
- Bornstein, R. (2010) Succession planning for the higher education presidency. AGB Press.
- Bozyigit, E. (2019) 'The Importance of Leadership Education in University: Self-Leadership Example', *International Education Studies*, 12(4), p. 1.
- Breckenridge, J. P., Jones, D., Elliott, I. and Nicol, M. (2012) 'Choosing a Methodological Path: Reflections on the Constructivist Turn', *The Grounded Theory Review*, 11(1), pp. 64–71.
- Brugger, G. R. (1993) 'A study of a community college leadership succession event.', *The University of Utah*.
- Brundrett, M., Rhodes, C. and Gkolia, C. (2006) 'Planning for leadership succession: creating a talent pool in primary schools', *International Journal of Phytoremediation*, 34(3), pp. 259–268.
- Bryant, A. and Charmaz, K. (2007) *The Sage handbook of grounded theory, Sage.*Sage.

- Bush, T. (2011) 'Succession planning and leadership development for school principals: Comparing English and South African approaches', *Compare*, 41(6), pp. 785–800.
- Card, M. A. (1992) Creating Your Job: Leadership, Executive Succession, and Strategy Formation by State Agency Directors in Ohio, The Ohio State University.
- Carey, D. and Ogden, D. (2000) CEO succession, Oxford University Press.
- Carlson, K. (2007) Succession planning: Identifying and preparing future leaders in the Colorado Community College System 2007,.
- Cavanaugh, J. (2017) 'Who Will Lead? The Success of Succession Planning', Journal of Management Policy and Practice, 18(2), pp. 22–27.
- Chang, C. and Besel, K. (2020) 'Cultivating Next Generation of Healthcare Leaders in Havana: Barriers and Recommendation for Succession Planning', *Journal of Healthcare Management*, 14(4), pp. 1062–1070.
- Channell, T. L. (2013) 'Presidential Succession Planning for Governing Boards in Higher Education'.
- Charmaz, K. (2006) Constructing grounded theory: A practical guide through qualitative analysis, SAGE Publications. Sage.
- Charmaz, K. (2008) 'Grounded theory as an emergent method. In S. N. Hesse-Biber, & P.Leavy (Eds.)', in *Handbook of Emergent Method*, pp. 155–170.
- Charmaz, K. (2014) Constructing grounded theory, Sage. Sage.
- Chesney-Lind, M. and Faith, K. (2001) 'What about feminism? Engendering theorymaking in criminology', in *Explaining criminals and crime*, pp. 287–302.
- Cimirotić, R., Duller, V., Feldbauer-Durstmüller, B., Gärtner, B. and Hiebl, M. R. W. (2017) 'Enabling factors that contribute to women reaching leadership positions in business organizations: The case of management accountants', *Management Research Review*, 40(2), pp. 165–194.
- Clark, V. L. P. and Creswell, J. W. (2015) *Understanding Research A Consumer's Guide*.
- Clunies, J. (2004) 'Benchmarking Succession Planning & Executive Development in Higher Education', *Academic Leadership Journal*, 2(4).
- Cobanoglu, N. (2020) 'Investigation of Shared Leadership and Organizational Commitment in Primary and Secondary Schools: Malatya Case', *International Journal of Educational Methodology*, 6(3), pp. 613–629.

- Cole, K., Hassel, H. and Schell, E. E. (2017) 'Feminist Decision Making and Resistance to Academic Neoliberalism', in *Surviving Sexism in Academia*. Routledge, pp. 13–28.
- Conger, J. A. and Fulmer, R. M. (2003) 'Developing your leadership pipeline.', *Harvard Business Review*, 81(12), pp. 76–84.
- Cooper, T. (2019) 'Rethinking teaching excellence in Australian higher education', International Journal of Comparative Education and Development, 21(2), pp. 83–98.
- Cortes, G. M., Jaimovich, N. and Siu, H. E. (2018) The 'End of Men' and Rise of Women in the High-Skilled Labor Market, National Bureau of Economic Research.
- Creswell, J. W. (2012) Educational research: Planning, Conducting, and Evaluating

 Ouantitative and Oualitative Research, Pearson.
- Creswell, J. W. and Clark, V. L. P. (2007) Designing and Conducting Mixed Methods Research [Book Review], Sage publications.
- Currie, J., Thiele, B. and Harris, P. (2002) Gendered universities in globalized economies: Power, careers, and sacrifices, Lexington Books. Lexington Books.
- Dahlvig, J. E. and Longman, K. A. (2020) 'Women's leadership in higher education: Addressing glass ceilings and sticky floors', *The Wiley Handbook of Gender Equity in Higher Education*, pp. 29–52.
- Darvish, H. and Najafi, Z. (2014) 'A Study on the Relationship between Succession Planning and Strategic Planning. Case Study: Payame Noor University of Aleshtar', *Economic Insights Trends & Challenges*, 64(1), pp. 11–24.
- David, M. E. (2015) 'Women and gender equality in higher education?', *Education Sciences*, 5(1), pp. 10–25.
- Davis, E. B. (2008) 'Colleges need to offer clear paths to leadership', *The Chronicle of Higher Education*, 54(45).
- Denzin, N. K. and Lincoln, Y. S. (2005) The Sage Handbook of Qualitative Research.
- Desarno, J., Perez, M., Rivas, R., Sandate, I., Reed, C. and Fonseca, I. (2021) 'Succession Planning Within the Health Care Organization:: Human Resources Management and Human Capital Management Considerations', *Nurse Leader*. Elsevier Inc., 19(4), pp. 411–415.

- Diehl, A. B. and Dzubinski, L. M. (2016) 'Antecedents and Outcomes of Workplace Incivility', *Human Resource Development Quarterly*, 27(2), pp. 181–206.
- Diezmann, C. M. and Grieshaber, S. J. (2010) 'Gender equity in the professoriate: A cohort study of new women professors in Australia', in *HERDSA Conference*.
- Dopson, S., Ferlie, E., McGivern, G., Fischer, M. D., Mitra, M., Ledger, J. and Behrens, S. (2018) 'Leadership Development in Higher Education: A Literature Review and Implications for Programme Redesign', *Higher Education Quarterly*, 73(2), pp. 218–234.
- Du-Plessis, M. A. (2019) a Succession Planning Framework for Leadership Development of Nurse Educators in Nursing Education Institutions in South Africa.
- Ebbers, L., Conover, K. S. and Samuels, A. (2010) 'Leading from the Middle: Preparing Leaders for New Roles', *New Directions for Community Colleges*, (149), pp. 59–64.
- Eisenhardt, K. M. (1989) 'Building Theories from Case Study Research.', *Academy of Management Review*, 14(4), pp. 532–550.
- El-Alayli, A., Hansen-Brown, A. A. and Ceynar, M. (2018) 'Dancing Backwards in High Heels: Female Professors Experience More Work Demands and Special Favor Requests, Particularly from Academically Entitled Students', *Sex Roles*. Sex Roles, 79(3–4), pp. 136–150.
- Elamin, A. M. and Omair, K. (2010) 'Males' Attitudes Towards Working Females in Saudi Arabia', *Personnel Review*, 39(6), pp. 746–766.
- Fallatah, R. and Syed, J. (2017) Employee Motivation in Saudi Arabia: An Investigation into the Higher Education Sector, Springer.
- Fulton-calkins, P. and Milling, C. (2010) 'Community College Journal of COMMUNITY-COLLEGE LEADERSHIP: AN ART TO BE PRACTICED: 2010 AND BEYOND', (February 2015), pp. 37–41.
- Fusarelli, B. C., Fusarelli, L. D. and Riddick, F. (2018) 'Planning for the Future: Leadership Development and Succession Planning in Education', *Journal of Research on Leadership Education*, 13(3), pp. 286–313.
- Gaffney, S. (2005) 'Career Development as a Retention and Succession Planning Tool.', *The Journal for Quality and Participation*, 28(3), p. 7.

- Gandhi, M. and Sen, K. (2021) 'Missing women in Indian university leadership:

 Barriers and facilitators', *Educational Management Administration and Leadership*, 49(2), pp. 352–369.
- Gangone, L. and Lennon, T. (2014) 'Benchmarking women's leadership in academia and beyond', in *Women and leadership in higher education*, pp. 3–22.
- Garland, B. (2001) 'The Role of the Universities and CANTEACH in Succession Planning'.
- Gilbert, S. A. (2017) Succession Planning Relating to The Millennial Generation in Private Four-Year Universities.
- Glaser, B. G. (1978) Advances in the methodology of grounded theory: Theoretical sensitivity, University of California.
- Glaser, B. G. (1992) Basics of grounded theory analysis. Mill Valley.
- Glaser, B. G. (1998) Doing grounded theory: Issues and discussions. Sociology Press.
- Glaser, B. G. and Strauss, A. L. (1967) *Grounded theory: Strategies for qualitative research*. Chicago, lL: Aldine Publishing Company.
- Gola, C. H. and Bennett, M. H. (2016) 'The University of Houston's Liaison Services Advisory Board: A case study in leadership development and succession planning', *College and Research Libraries News*, 77(10), pp. 496–501.
- Golafshani, N. (2003) 'Understanding Reliability and Validity in Qualitative Research', *The Qualitative Report*, 8(4), pp. 597–606.
- Goldman, C. (2020) 'Structure, Culture, and Agency: Examining Succession Planning in California State University (CSU) Libraries', *Journal of Library Administration*. Routledge, 60(1), pp. 1–21.
- Goulding, C. (1998) 'Grounded Theory: The Missing Methodology on the Interpretivist Agenda', *Qualitative Market Research: An International Journal*, 1(1), pp. 50–57.
- Grigoli, F., Koczan, Z. and Topalova, P. (2018) A Cohort-Based Analysis of Labor Force Participation for Advanced Economies.
- Guarino, C. M. and Borden, V. M. H. (2017) 'Faculty Service Loads and Gender: Are Women Taking Care of the Academic Family?', *Research in Higher Education*, 58(6), pp. 672–694.

- Hall-Ellis, S. D. and Grealy, D. S. (2013) 'The Dreyfus model of skill acquisition: A career development framework for succession planning and management in academic libraries', *College and Research Libraries*, 74(6), pp. 587–603.
- Hannon, P. (2014) Succession Planning: Developing The Next Generation of Nurse Leaders, Texas Woman's University.
- Harford, J. (2018) 'The perspectives of women professors on the professoriate: A missing piece in the narrative on gender equality in the university', *Education Sciences*, 8(2), pp. 1–17.
- Hartley III, H. V. and Godin, E. E. (2009) 'A Study of Career Patterns of the Presidents of Independent Colleges and Universities.', *Council of Independent Colleges*, p. 36.
- Heath, H. and Cowley, S. (2004) 'Developing a grounded theory approach: a comparison of Glaser and Strauss', 41, pp. 141–150.
- Hinojosa, J. L. (2019) 'The Athenaeum A Grounded Theory Study of Senior Leader Experiences and Perceptions of Unplanned Turnover'.
- Hodges, j. (2017) 'Cracking the walls of leadership: women in Saudi Arabia', Gender in management: an international journal, 32(1), pp. 34–46.
- Holton, J. A. (2010) 'The Coding Process and Its Challenges Judith', *The Grounded Theory Review*, 9(1), pp. 21–40.
- Howe-Walsh, L. and Turnbull, S. (2016) 'Barriers to women leaders in academia: tales from science and technology', *Studies in Higher Education*, 41(3), pp. 415–428.
- Hunt, E. A. (2009) 'Minding the Mission: Succession Planning for Academic Leadership at Christian Institutions'.
- Hunt, J. G. (Jerry) and Ropo, A. (1995) 'Multi-level Leadership: Grounded Theory and Mainstream Theory Applied to the Case of General Motors', *Leadership Quarterly*, 6(3), pp. 379–412.
- Hutchings, K., Lirio, P. and Metcalfe, B. D. (2012) 'Gender, globalisation and development: A re-evaluation of the nature of women's global work', *International Journal of Human Resource Management*, 23(9), pp. 1763– 1787.
- Ishak, A. K. and Kamil, B. A. M. (2016) 'Succession planning at higher education institutions: Leadership style, career development and knowledge

- management practices as its predictors', *International Review of Management and Marketing*, 6(7Special Issue), pp. 214–220.
- Jackson, N. C. and Dunn-Jensen, L. M. (2021) 'Leadership succession planning for today's digital transformation economy: Key factors to build for competency and innovation', *Business Horizons*. Elsevier Ltd, 64(2), pp. 273–284.
- Jacobs, J. A. and Winslow, S. E. (2004) 'The academic life course, time pressures and gender inequality', *Community, Work and Family*, 7(2), pp. 143–161.
- Jamjoom, F. B. and Kelly, P. (2013) 'Higher Education for Women in the Kingdom of Saudi Arabia Women 's Education in Saudi Society: A Brief Overview', pp. 117–125.
- Jones, P. and Rivers, R. (2021) 'Development of a Staff Nurse to Chief Nursing Officer Succession Planning Program', *Nurse Leader*. Elsevier Inc., 19(6), pp. 646–654.
- Kairys, M. R. (2018) 'The influence of gender on leadership in education management', *International Journal of Educational Management*, 32(5), pp. 931–941.
- Kallio, K. M., Kallio, T. J., Tienari, J. and Hyvönen, T. (2016) 'Ethos at stake: Performance management and academic work in universities', *Human Relations*, 69(3), pp. 685–709.
- Karam, C. M. and Afiouni, F. (2014) 'Localizing women's experiences in academia: multilevel factors at play in the Arab Middle East and North Africa', *International Journal of Human Resource Management*, 25(4), pp. 500–538.
- Kattan, M. M. (2015) Factors of Successful Women Leadership in Saudi Arabia.
- Keddy, B., Sims, S. L. and Stern, P. N. (1996) 'Grounded Theory as Feminist Research Methodology', *Journal of Advanced Nursing*, 23(3), pp. 448–453.
- Keller, K. (2018) 'Building the Case for Succession Planning in Higher Education: A Study of Succession Planning Pilots within the Minnesota State Colleges and Universities ...'.
- Kemp, L. J., Madsen, S. R. and El- Saidi, M. (2013) 'The current state of female leadership in the United Arab Emirates', *Journal of Global Responsibility*, 4(1), pp. 99–112.
- Kezar, A. J. (2017) 'Leadership in Higher Education, Concepts and Theories', Encyclopedia of International Higher Education Systems and Institutions.

- King, S. R. (2019) Succession Planning and Leadership Development In Texas Public Universities.
- Kirkland, T. P. (1990) A grounded theory of community college chief executive officer succession.
- Kitchenham, B., Pearl Brereton, O., Budgen, D., Turner, M., Bailey, J. and Linkman, S. (2009) 'Systematic Literature Reviews in Software Engineering A Systematic Literature Review', *Information and Software Technology*. Elsevier B.V., 51(1), pp. 7–15.
- Klein, M. F. and Salk, R. J. (2013) 'Presidential succession planning: A qualitative study in private higher education', *Journal of Leadership and Organizational Studies*, 20(3), pp. 335–345.
- Kocolowski, M. (2010) 'Shared leadership: Is it time for a change', *Emerging Leadership Journeys*, 3(1), pp. 22–32.
- Komives, S. R., Owen, J. E., Longerbeam, S. D., Mainella, F. C. and Osteen, L. (2005) 'Developing a leadership identity: A grounded theory', *Journal of College Student Development*, 46(6), pp. 593–611.
- Land, P. C. (2003) 'From the Other Side of the Academy to Academic Leadership Roles: Crossing the Great Divide', (124), pp. 13–20.
- Lapovsky, L. (2007) 'The Best-Laid Succession Plans.', Association of Governing Boards of Universities and Colleges, pp. 14–1.
- Larkin, J. and Neumann, R. (2012) 'Ageing Academics: Workforce Priorities for Universities', 20(1), pp. 3–24.
- Lerner, G. (1986) The creation of patriarchy (Vol.1), Women and History.
- Lin, M. (2019) 'Educational Thoughts of Western Radical Feminism', 15(3), pp. 21–25.
- Longman, K. A. and Anderson, P. S. (2016) 'Women in Leadership: The Future of Christian Higher Education', *Christian Higher Education*. Taylor & Francis, 15(1–2), pp. 24–37.
- Loomes, S., Owens, A. and McCarthy, G. (2019) 'Patterns of recruitment of academic leaders to Australian universities and implications for the future of higher education', *Journal of Higher Education Policy and Management*, 41(2), pp. 137–152.

- Lu, J., Laux, C. and Antony, J. (2017) 'Lean Six Sigma leadership in higher education institutions', *International Journal of Productivity and Performance Management*, 66(5), pp. 638–650.
- Luna, G. (2010) 'Succession planning: A doctoral program partnership for emerging community college leaders', *Community College Journal of Research and Practice*, 34(12), pp. 977–990.
- Luzebetak, A. K. (2010) 'Community college succession planning: Preparing the next generation of women for leadership roles.', *Dissertation Abstracts International Section A: Humanities and Social Sciences*, 71, p. 1885.
- Macfarlane, B. and Burg, D. (2018) 'Women Professors as Intellectual Leaders', in *Project report, Leadership Foundation for Higher Education*, pp. 1–40.
- Magner, D. K. (2009) 'The Leadership Pipeline', Chronicle of Higher Education.
- Mapolisa, T., Mhlanga, E., Chabvonga, N. and Chimbwadzwa, Z. (2013) 'Female Leadership Dilemmas in Primary Schools: A Case Study of Primary Schools in Harare Province in Zimbabwe', *Zimbabwe Journal of Educational Research*, 25(2), pp. 261–276.
- Martin, V. B. and Gynnild, A. (2011) *Grounded theory: The philosophy, method, and work of Barney Glaser*, 2011. Universal-Publishers.
- Mateso, P. E. . E. (2010) 'Understanding succession planning and management efforts at Midwestern University: a mixed methods study', p. 270.
- Mauthner, N. S. and Edwards, R. (2010) 'Feminist research management in higher education in Britain: Possibilities and practices', *Gender, Work and Organization*, 17(5), pp. 481–502.
- McMurray, A. M., Henly, D., Chaboyer, W., Clapton, J., Lizzio, A. and Teml, M. (2012) 'Leadership succession management in a university health faculty', *Journal of Higher Education Policy and Management*, 34(4), pp. 365–376.
- Meek, V. (2007) 'Succession planning and Australian higher education', in *The Treasure Within Leadership and Succession Planning*, pp. 111–125.
- Melton, L. C. (2018) 'Succession planning research in the educational sector', Succession Planning: Promoting Organizational Sustainability, pp. 175–186.
- Meriam, sharan b (2009) Qualitative Research A Guide to Design and Implementation Revised.

- Metcalfe, A. and Slaughter, S. (2008) 'The differential effects of academic capitalism on women in the academy', in *Unfinished agendas: New and continuing gender challenges in higher education*, pp. 80–111.
- Mifsud, D. (2019a) 'Introduction: The Positioning and Making of Female Academics—A Review of the Literature', in *The Positioning and Making of Female Professors: Pushing Career Advancement Open*, pp. 1–26.
- Mifsud, D. (2019b) 'Introduction: The Positioning and Making of Female Academics—A Review of the Literature', *Springer*.
- Miles, M. B., Huberman, M. a and Saldana, J. (2014) 'Drawing and Verying Conclusions', *Qualitative Data Analysis: A Methods Sourcebook*, pp. 275–322.
- Miller, J. (2010) 'Commentary on heidensohn's "the deviance of women": Continuity and change over four decades of research on gender, crime and social control', *British Journal of Sociology*, 61(SUPPL. 1), pp. 133–139.
- Miller, J. and Mullins, C. W. (2006) 'The Status of Feminist Theories in Criminology', in *Taking Stock*. Routledge, pp. 217–249.
- Míltersteíner, R. K., de Oliveira, F. B., Hryniewicz, L. G. C., Santanna, A. de S. and Moura, L. C. (2020) 'Female leadership: perceptions, reflections, and challenges in public administration', *Cadernos EBAPE*, 18(2), pp. 406–423.
- Mohnot, H. (2017) 'The impact of some demographic variables on academic leadership preparedness in Indian higher education', *Journal of Further and Higher Education*. Routledge, 9486(August), pp. 1–11.
- Morley, L. (2014) 'Lost leaders: Women in the global academy', *Higher Education Research and Development*, 33(1), pp. 114–128.
- Morrin, A. (2013) 'Factors that are important to succession planning: A case study of one Ontario college of applied arts and technology', *College Quarterly*, 16(1), pp. 1–18.
- Morse, J. M., Bowers, B. J., Charmaz, K., Corbin, J., Clarke, A. E. and Stern, P. N. (2016) *Developing grounded theory: The second generation, Routledge*.
- Moser, K. (2008) 'California's Community Colleges Struggle to Recruit and Retain President', *Chronicle of Higher Education*, 55(13).
- Moss-Racusin, C. A., Dovidio, J. F., Brescoll, V. L., Graham, M. J. and Handelsman, J. (2012) 'Science faculty's subtle gender biases favor male students',

- Proceedings of the National Academy of Sciences of the United States of America, 109(41), pp. 16474–16479.
- Mumford, M. D. (2000) 'Managing Creative People: Strategies and Tactics for Innovation', *Human Resource Management Review*, 10(3), pp. 313–351.
- Murray, R. and Mifsud, D. (2019) *The positioning and making of female professors:*Pushing career advancement open, Palgrave macmilla.
- Mustafa, R. F. and Troudi, S. (2019) 'Saudi Arabia and Saudi Women in Research Literature: A Critical Look', *Asian Social Science*, 15(2), p. 133.
- Ndunguru, G. (2018) 'The practice of Succession Planning in Tanzania's Higher Learning Institutions a Case of Local Government Training Institute (LGTI)', p. 115.
- Neefe, D. O. (2009) 'Succession Planning in a Two-Year Technical College System',

 A Dissertation Submitted to the Faculty of the Graduate School of the

 University of Minnesota, (May).
- Negrea, S. (2008) 'Succession planning: A necessity for community colleges', Institute for Community College Development.
- Nielsen, T. H. (1987) 'The Creation of Patriarchy', WOMEN AND HISTORY-LERNER, 1, pp. 292–232.
- Ollilainen, M. and Solomon, C. R. (2014) 'Carving a "Third Path": Faculty Parents' Resistance to the Ideal Academic Worker Norm', *Advances in Gender Research*, 19, pp. 21–39.
- Olson-Strom, S. and Rao, N. (2020) 'Higher Education for Women in Asia', in *Diversity and Inclusion in Higher Education: Lessons from accross Asia*, pp. 263–282.
- Omar, R. and Hassan, Z. A. (2015) 'Blooming Leaders Among Equals: a Case of Leadership Succession Planning in Malaysian Higher Education', *Scientific Cooperations International Journal of Finance Business, Economics, Marketting and Information Systems*, 1(1), pp. 30–40.
- Owolabi, T. and Adeosun, O. (2021) 'Succession Planning and Talent Retention: Evidence From the Manufacturing Sector in Nigeria', *British Journal of Management*, 4(January), pp. 17–32.
- Parkman, A. and Beard, R. (2008) 'Succession planning and the imposter phenomenon in higher education', *CUPA-HR Journal*, 59(2), pp. 29–36.

- Parry, K. W. (1998) 'Grounded Theory and Social Process: A New Direction for Leadership Research', *Leadership Quarterly*, 9(1), pp. 85–105.
- Pate, S. S. and Adams, M. (2013) 'The Influence of Social Networking Sites on Buying Behaviors of Millennials Part of the Advertising and Promotion Management Commons, Fashion Business Commons, Marketing Commons, and the Sales and Merchandising Commons', *Atlantic Marketing Journal*, 2(1), pp. 4–27.
- Patton, M. Q. (2002) *Qualitative research & evaluation methods (3rd ed.)*. Thousand Sage: Oaks.
- Petticrew, M. and Roberts, H. (2008) Systematic Reviews in the Social Sciences: A Practical Guide.
- Phillips, L. K. (2021) 'Succession planning in nursing education', *Nursing Outlook*. Elsevier Inc., 69(1), pp. 32–42.
- Pierce, D. (2015) 'Guiding light. How community college presidents can ensure their vision lives on, even after they move on', *Community College Journal*, 85(6), pp. 26–32.
- Power, A. (2020) 'Understanding leadership in higher education as a tool for change in relation to gender', in *Gender in academic recruitment and selection*, pp. 140–153.
- Probert, B. (2005) "I Just Couldn't Fit It In": Gender and Unequal Outcomes in Academic Careers', *Gender, Work and Organization*, 12(1), pp. 50–72.
- Raby, R. L. and Valeau, E. J. (2019) 'Position Training and Succession Planning for Community College International Education Leaders', *Community College Journal of Research and Practice*. Routledge, 45(2), pp. 86–102.
- Rich, D. (2006) 'Academic leadership and the restructuring of higher education', *New Directions for Higher Education*, 2006(134), pp. 37–48.
- Richards, C. L. (2009) A New Paradigm: Strategies For Succession Planning In Higher Education, Capella University.
- Richardsen, A. M., Traavik, L. E. M. and Burke, R. J. (2016) 'Women and work stress: more and different?', in *Handbook on Well-Being of Working Women*, pp. 123–140.
- Ritchie, M. (2020) 'Succession planning for successful leadership: Why we need to talk about succession planning!', *Management in Education*, 34(1), pp. 33–37.

- Röbken, H. (2007) 'Leadership Turnover Among University Presidents', *Management Revue*, 18(2), pp. 138–152.
- Rosenthal, J., Routch, K., Monahan, K. and Doherty, M. (2018) 'The holy grail of effective leadership succession planning: How to overcome the succession planning paradox', *Deloitte Insights*.
- Rothwell, W. (2010) Effective succession planning: Ensuring leadership continuity and building talent from within (4th ed.), Amacom. Amacom.
- Salari, S. and Abedini, M. (2017) 'Examining the role of management succession planning (MSP) in talent identification and empowerment of human capital in Hormozgan University of Medical Sciences', *Hormozgan Medical Journal*, 21(2), pp. 119–128.
- Saldana, J. (2011) Fundamentals of Qualitative Research.
- Santovec, M. Lou (2010) 'Succession Planning Gives Women a Shot at Leadership', *Women in Higher Education*. John Wiley & Sons, Ltd, 19(5), pp. 6–7.
- Saunders, B. and Kitzinger, J. (2015) 'Anonymising interview data: challenges and compromise in practice'.
- Savigny, H. (2014) 'Women, Know Your Limits: Cultural Sexism in Academia', in *Gender and Education*, pp. 794–809.
- Schueller-Weidekamm, C. and Kautzky-Willer, A. (2012) 'Challenges of work-life balance for women physicians/mothers working in leadership positions', *Gender Medicine*. Excerpta Medica, Inc., 9(4), pp. 244–250.
- Scott, H. M. P. (2007) The Temporal Integration of Connected Study into a Structured Life: A Grounded Theory.
- Semela, T., Bekele, H. and Abraham, R. (2020) 'Navigating the river Nile: the chronicle of female academics in Ethiopian higher education', *Gender and Education*. Taylor & Francis, 32(3), pp. 328–346.
- Shamsuddin, A., Chan, C., Sari, A. and Kassim, M. (2012) 'Leadership Management as an Integral Part of Succession Planning in HEIs: A Malaysian Perspective', *International Journal of Business and Social Science*, 3(3), pp. 151–159.
- Shava, G. N. and Tlou, F. N. (2018) 'Distributed Leadership in Education , Contemporary Issues in Educational Leadership', *African Educational Research Journal*, 6(4), pp. 279–287.

- Shaw, S. and Hernandez-gantes, V. (2021) 'The Status of Female Executive Leadership in Postsecondary Education', 32, pp. 1–16.
- Sherman, R., Dyess, S., Hannah, E. and Prestia, A. (2013) 'Succession planning for the future through an academic-practice partnership: A nursing administration master's program for emerging nurse leaders', *Nursing Administration Quarterly*, 37(1), pp. 18–27.
- Shults, C. (2001) 'The Critical Impact of Impending Retirements on Community College Leadership.', American Association of Community Colleges, Washington, DC.
- Sidani, Y. and Al Ariss, A. (2014) 'Institutional and Corporate Drivers of Global Talent Management: Evidence from the Arab Gulf Region', *Journal of World Business jo*, 49, pp. 215–224.
- Sidani, Y. M., Konrad, A. and Karam, C. M. (2015) 'From female leadership advantage to female leadership deficit A developing country perspective', *Career Development International*, 20(3), pp. 273–292.
- Simmons, J. A. (2009) "Both sides now": Aligning external and internal branding for a socially responsible era', *Marketing Intelligence and Planning*, 27(5), pp. 681–697.
- Song, W. and Hartley III, H. V. (2012) 'A study of presidents of independent colleges and universities', pp. 1–48.
- Spanier, G. B. (2010) 'The president as mentor.', AGB Trusteeship, 18(3).
- Stern, P. N. (2009) 'In the beginning Glaser and Strauss created grounded theory', in *Developing grounded theory: The second generation*, pp. 23–29.
- Stevens-Kalceff, M., Hagon, S., Cunningham, M. and Woo, A. (2007) *Maximising Potential in Physics: Investigation of the Academic Profile of the School of Physics at the University of New South Wales*.
- Straus, A. and Corbin, J. (1994) 'Grounded theory methodology: An overview', in *Handbook of qualitative research*, pp. 273–285.
- Stripling, J. and Medina, B. (2011) 'The graying presidency.', *The Chronicle of Higher Education*, 58(6), pp. 1–10.
- Swanson, M. A. (2018) 'Employee Perceptions of Succession Planning Within Higher Education a Qualitative Case Study', (May).

- Syed, J., Ali, F. and Hennekam, S. (2018) 'Gender equality in employment in Saudi Arabia: a relational perspective', *Career Development International*, 23(2), pp. 163–177.
- Tichy, N. M. (2014) Succession: Mastering the make or break process of leadership transition, go.gale.com. Penguin.
- Till, A., McKimm, J. and Swanwick, T. (2020) 'The importance of leadership development in medical curricula: A UK perspective (stars are aligning)', *Journal of Healthcare Leadership*, 12, pp. 19–25.
- Tlaiss, H. and Kauser, S. (2011) 'The Impact of Gender, Family, and Work on the Career Advancement of Lebanese Women Managers', *Gender in Management: An International Journal*, 26(1), pp. 8–36.
- Torabi, N., Karimi, F. and Nadi, M. A. (2019) 'Strategies For Succession Planning In Universities: A Mixed-Methods Analysis', *MALAYSIAN ONLINE JOURNAL OF EDUCATIONAL MANAGEMENT (MOJEM)*, 7(2), pp. 46–71.
- Tran, T. Q. (2020) 'Identifying female leadership and performance in small and medium-sized enterprises in a transition economy: The case study of Vietnam', *Asian Economic and Financial Review*, 10(2), pp. 132–145.
- Tranfield, D., Denyer, D. and Smart, P. (2003) 'Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review', *British Journal of Management*, 14, pp. 207–222.
- Turner, P. (2018) 'The Role of Leaders: The Importance of Leadership', *Leadership in Healthcare*, (Yukl 2010), pp. 45–74.
- UNDP (2020) '2020 HDR Technical Notes', pp. 1–19.
- Varshney, D. (2019) 'The Strides of the Saudi female workforce: Overcoming constraints and contradictions in transition', *Journal of International Women's Studies*, 20(2), pp. 359–372.
- Walker, D. and Myrick, F. (2006) 'Qualitative Health Research', *Qualitative health research*, 16(4), pp. 547–559.
- Washington, C. E. (2016) 'Succession planning in higher education'.
- De Welde, K., Ollilainen, M. and Solomon, C. R. (2019) 'Feminist Leadership in the Academy: Exploring Everyday Praxis', *Advances in Gender Research*, 27, pp. 3–21.
- De Welde, K. and Stepnick, A. (2015) 'From People to Policies: Enduring Inequalities and Inequities for Women Academics', in *Disrupting the culture*

- of silence: Confronting gendDe Welde, K. and Stepnick, A. (2015) 'From People to Policies: Enduring Inequalities and Inequities for Women Academics', in Disrupting the culture of silence: Confronting gender inequality and making cha, pp. 1–28.
- Wilkins, S. and Neri, S. (2019) 'Managing Faculty in Transnational Higher Education: Expatriate Academics at International Branch Campuses', *Journal of Studies in International Education*, 23(4), pp. 451–472.
- World Economic Forum (2018) 2018 The Global Gender Gap Report, World Economic Forum.
- Wright, R. T., Campbell, D. E., Thatcher, J. B., Roberts, N., Wright, R. T. and Roberts, N. (2012) 'Operationalizing Multidimensional Constructs in Structural Equation Modeling: Recommendations for IS Research Modeling: Recommendations for IS Research', 30.
- Wu, Q., Cormican, K. and Chen, G. (2020) 'A Meta-Analysis of Shared Leadership: Antecedents, Consequences, and Moderators', *Journal of Leadership and Organizational Studies*, 27(1), pp. 49–64.
- Yin, R. K. (2003) 'Designing case studies', *Qualitative research methods*, 5(14), pp. 359–386.
- Abalkhail, J. M. (2017) 'Women and leadership: challenges and opportunities in Saudi higher education', *Career Development International*, 22(2), pp. 165–183.
- Abalkhail, J. M. and Allan, B. (2015) 'Women's career advancement: Mentoring and networking in Saudi Arabia and the UK', *Human Resource Development International*, 18(2), pp. 153–168.
- Abdullah, Z., Samah, S. A. A., Jusoff, K. and Isa, P. M. (2009) 'Succession Planning in Malaysian Institution of Higher Education', *International Education Studies*, 2(1), pp. 2005–2008.
- Acker, S. and Wagner, A. (2019) 'Feminist scholars working around the neoliberal university', *Gender and Education*, 31(1), pp. 62–81.
- Adewale, O., Abolaji, A. and Kolade, O. (2011) 'Succession planning and organizational survival: Empirical study on Nigerian private tertiary institutions', *Serbian Journal of Management*, 6(2), pp. 231–246.
- Akram, F., Murugiah, L. a/p and Shahzad, A. (2017) 'Cultural Aspects and Leadership Effectiveness of Women Leaders: A Theoretical prospective of

- Saudi Arabia', *Pakistan Journal of Humanities and Social Sciences*, 5(1), pp. 25–35.
- Al-Asfour, A., Tlaiss, H. A., Khan, S. A. and Rajasekar, J. (2017) 'Saudi women's work challenges and barriers to career advancement', *Career Development International*, 22(2), pp. 184–199.
- Al-bakr, F., Bruce, E., Davidson, P., Schlaffer, E. and Kropiunigg, U. (2017) 'Empowered but not Equal: Challenging the Traditional Gender Roles as Seen by University Students in Saudi Arabia', *Forum For International Research in Education*, 4(1), pp. 52–66.
- Al-jaradat, M. K. M. (2014) 'Challenges Facing Women Academic Leadership in Secondary Schools of Irbid Educational Area', 7(5).
- Al-Ohali, M. and Al-Mehrej, H. (2012) 'Faculty salary and remuneration in the Kingdom of Saudi Arabia.', in *Paying the professoriate: A global comparison of compensation and contracts*, pp. 278–287.
- Al-Sudairy, H. (2017) Modern woman in the Kingdom of Saudi Arabia: Rights, challenges and achievements, Cambridge Scholars Publishing. Cambridge Scholars Publishing.
- Albakry, A. H. (2016) Voices of women leaders in corporate institutions in Saudi Arabia.
- Alexander, N. H. (2011) 'Teaching Leadership to Female Students in Saudi Arabia', Advancing Women in Leadership, 31, pp. 199–212.
- Alexander, N. H. (2013) 'Saudi Arabia Female Students' Perception of Effective Female Leaders.', *Advancing Women in Leadership*, 33, pp. 142–150.
- Alghofaily, L. (2019) Investigating the Perceived Barriers to Women's Rise to Leadership in Higher Education in Saudi Arabia.
- Al Alhareth, Yahya, Al Alhareth, Yasra and Al Dighrir, I. (2015) 'Review of Women and Society in Saudi Arabia', *American Journal of Educational Research*, 3(2), pp. 121–125.
- Ali, Z. and Mehreen, A. (2019) 'Understanding succession planning as a combating strategy for turnover intentions', *Journal of Advances in Management Research*, 16(2), pp. 216–233.
- Alimo-metcalfe, B. and Alimo-metcalfe, B. (2010) 'An investigation of female and male constructs of leadership and empowerment'.

- Alomair, M. O. (2015) 'Female Leadership Capacity and Effectiveness: A Critical Analysis of the Literature on Higher Education in Saudi Arabia', *International Journal of Higher Education*, 4(4), pp. 81–93.
- Alotaibi, F., Cutting, R. and Morgan, J. (2017) 'A Critical Analysis of the Literature in Women's Leadership in Saudi Arabia', *International Journal of Business Administration and Management Research*, 3(1), p. 29.
- Alsubaie, A. and Jones, K. (2017) 'An Overview of the Current State of Women's Leadership in Higher Education in Saudi Arabia and a Proposal for Future Research Directions', *Administrative Sciences*, 7(4), p. 36.
- Anderson, E. K. and Solomon, C. R. (2015) Family-friendly policies and practices in academe 2015,.
- Anderson, M. (2017) 'Transformational Leadership in Education: A Review of Existing Literature', *International Social Science Review*, 93(1), p. 4.
- Ayalew, S., Manian, S. and Sheth, K. (2021) 'Discrimination from below: Experimental evidence from Ethiopia', *Journal of Development Economics*, 151, pp. 1–37.
- Ayandibu, A. O. and Kaseeram, I. (2020) 'The Future of Workforce Planning', in *igi-global.com*, pp. 127–156.
- Aylett, A. (2019) 'Influencers of Succession Planning Among Rural Community College Chief Executives', *ProQuest Dissertations and Theses*, p. 97.
- Barden, D. M. (2010) 'Where will you find your future leaders', *Chronicle of Higher Education*, 56(17).
- Barginere, C., Franco, S. and Wallace, L. (2013) 'Succession planning in an academic medical center nursing service', *Nursing Administration Quarterly*, 37(1), pp. 67–71.
- Barnett, R. and Davis, S. (2008) 'Creating Greater Success in Succession Planning', *Advances in Developing Human Resources*, 10(5), pp. 721–739.
- Barraza Vargas, C. (2019) 'Women in Command: The Impact of Female Leadership on International Organisations', *Global Society*. Taylor & Francis, 33(4), pp. 541–564.
- Barton, A. (2019) 'Preparing for Leadership Turnover in Christian Higher Education:

 Best Practices in Succession Planning', *Christian Higher Education*.

 Routledge, 18(1–2), pp. 37–53.

- Barton, T. R. (2006) 'Feminist leadership: Building nurturing academic communities.', *Advancing Women in Leadership Journal*, 20.
- Beheshtifar, M. and Moghadam, M. N. (2011) 'Studying the competency-related models in succession planning', *European Journal of Economics, Finance and Administrative Sciences*, 34(34), pp. 113–121.
- Belknap, R. A. and Cruz, N. (2007) 'When I Was in My Home I Suffered a Lot: Mexican Women' s Descriptions of Abuse in Family of Origin', *Health care for women international*, 28(5), pp. 506–522.
- Bello, M. A. and Thompson, E. S. (2003) 'Recruiting the cataloguers for tomorrow: the need for succession planning in Ghanaian academic libraries', *Library Management*, 24(6), pp. 310–315.
- Berchelman, D. K. (2005) 'Succession planning', *The Journal for Quality and Participation*, 28(3), p. 11.
- Betts, Kristen and Betts, Keith (2009) 'Visible Administrative Career Paths ,
 Professional Development , Succession Planning & Commitment to
 Diversity', 7(2).
- Birks, M. and Mills, J. (2011) *Grounded theory: A practical guide*. Sage publications.
- Bisbee, D. C. and Miller, M. T. (2006) 'Executive succession practices in land grant universities', *Academic Leadership*, 4(3), p. 1.
- Black, S., Gardner, D., Pierce, J. and Steers, R. (2019) 'Talent Development and Succession Planning', *Human Resource Management*.
- Bleich, M. R. (2019) 'The professional development educator and leadership succession planning', *Journal of Continuing Education in Nursing*. Slack Incorporated, 50(3), pp. 104–106.
- Boerner, H. (2015) 'Game plan', Community College Journal, 85(6), p. 20.
- Boggs, G. R. and Kent, E. L. (2002) 'President Academy: An Evolution of Leadership Development', in *New directions for community colleges*, pp. 51–58.
- Bolden, R. and O'Regan, N. (2016) 'Digital Disruption and the Future of Leadership:

 An Interview With Rick Haythornthwaite, Chairman of Centrica and

 MasterCard', *Journal of Management Inquiry*, 25(4), pp. 438–446.
- Bornstein, R. (2010) Succession planning for the higher education presidency. AGB Press.

- Bozyigit, E. (2019) 'The Importance of Leadership Education in University: Self-Leadership Example', *International Education Studies*, 12(4), p. 1.
- Breckenridge, J. P., Jones, D., Elliott, I. and Nicol, M. (2012) 'Choosing a Methodological Path: Reflections on the Constructivist Turn', *The Grounded Theory Review*, 11(1), pp. 64–71.
- Brugger, G. R. (1993) 'A study of a community college leadership succession event.', *The University of Utah*.
- Brundrett, M., Rhodes, C. and Gkolia, C. (2006) 'Planning for leadership succession: creating a talent pool in primary schools', *International Journal of Phytoremediation*, 34(3), pp. 259–268.
- Bryant, A. and Charmaz, K. (2007) *The Sage handbook of grounded theory, Sage.*Sage.
- Bush, T. (2011) 'Succession planning and leadership development for school principals: Comparing English and South African approaches', *Compare*, 41(6), pp. 785–800.
- Card, M. A. (1992) Creating Your Job: Leadership, Executive Succession, and Strategy Formation by State Agency Directors in Ohio, The Ohio State University.
- Carey, D. and Ogden, D. (2000) CEO succession, Oxford University Press.
- Carlson, K. (2007) Succession planning: Identifying and preparing future leaders in the Colorado Community College System 2007,.
- Cavanaugh, J. (2017) 'Who Will Lead? The Success of Succession Planning', Journal of Management Policy and Practice, 18(2), pp. 22–27.
- Chang, C. and Besel, K. (2020) 'Cultivating Next Generation of Healthcare Leaders in Havana: Barriers and Recommendation for Succession Planning', *Journal of Healthcare Management*, 14(4), pp. 1062–1070.
- Channell, T. L. (2013) 'Presidential Succession Planning for Governing Boards in Higher Education'.
- Charmaz, K. (2006) Constructing grounded theory: A practical guide through qualitative analysis, SAGE Publications. Sage.
- Charmaz, K. (2008) 'Grounded theory as an emergent method. In S. N. Hesse-Biber, & P.Leavy (Eds.)', in *Handbook of Emergent Method*, pp. 155–170.
- Charmaz, K. (2014) Constructing grounded theory, Sage. Sage.

- Chesney-Lind, M. and Faith, K. (2001) 'What about feminism? Engendering theorymaking in criminology', in *Explaining criminals and crime*, pp. 287–302.
- Cimirotić, R., Duller, V., Feldbauer-Durstmüller, B., Gärtner, B. and Hiebl, M. R. W. (2017) 'Enabling factors that contribute to women reaching leadership positions in business organizations: The case of management accountants', *Management Research Review*, 40(2), pp. 165–194.
- Clark, V. L. P. and Creswell, J. W. (2015) Understanding Research A Consumer's Guide.
- Clunies, J. (2004) 'Benchmarking Succession Planning & Executive Development in Higher Education', *Academic Leadership Journal*, 2(4).
- Cobanoglu, N. (2020) 'Investigation of Shared Leadership and Organizational Commitment in Primary and Secondary Schools: Malatya Case', *International Journal of Educational Methodology*, 6(3), pp. 613–629.
- Cole, K., Hassel, H. and Schell, E. E. (2017) 'Feminist Decision Making and Resistance to Academic Neoliberalism', in *Surviving Sexism in Academia*. Routledge, pp. 13–28.
- Conger, J. A. and Fulmer, R. M. (2003) 'Developing your leadership pipeline.', *Harvard Business Review*, 81(12), pp. 76–84.
- Cooper, T. (2019) 'Rethinking teaching excellence in Australian higher education', International Journal of Comparative Education and Development, 21(2), pp. 83–98.
- Cortes, G. M., Jaimovich, N. and Siu, H. E. (2018) The 'End of Men' and Rise of Women in the High-Skilled Labor Market, National Bureau of Economic Research.
- Creswell, J. W. (2012) Educational research: Planning, Conducting, and Evaluating Quantitative and Qualitative Research, Pearson.
- Creswell, J. W. and Clark, V. L. P. (2007) Designing and Conducting Mixed Methods Research [Book Review], Sage publications.
- Currie, J., Thiele, B. and Harris, P. (2002) Gendered universities in globalized economies: Power, careers, and sacrifices, Lexington Books. Lexington Books.
- Dahlvig, J. E. and Longman, K. A. (2020) 'Women's leadership in higher education: Addressing glass ceilings and sticky floors', *The Wiley Handbook of Gender Equity in Higher Education*, pp. 29–52.

- Darvish, H. and Najafi, Z. (2014) 'A Study on the Relationship between Succession Planning and Strategic Planning. Case Study: Payame Noor University of Aleshtar', *Economic Insights Trends & Challenges*, 64(1), pp. 11–24.
- David, M. E. (2015) 'Women and gender equality in higher education?', *Education Sciences*, 5(1), pp. 10–25.
- Davis, E. B. (2008) 'Colleges need to offer clear paths to leadership', *The Chronicle of Higher Education*, 54(45).
- Denzin, N. K. and Lincoln, Y. S. (2005) The Sage Handbook of Qualitative Research.
- Desarno, J., Perez, M., Rivas, R., Sandate, I., Reed, C. and Fonseca, I. (2021) 'Succession Planning Within the Health Care Organization:: Human Resources Management and Human Capital Management Considerations', *Nurse Leader*. Elsevier Inc., 19(4), pp. 411–415.
- Diehl, A. B. and Dzubinski, L. M. (2016) 'Antecedents and Outcomes of Workplace Incivility', *Human Resource Development Quarterly*, 27(2), pp. 181–206.
- Diezmann, C. M. and Grieshaber, S. J. (2010) 'Gender equity in the professoriate: A cohort study of new women professors in Australia', in *HERDSA Conference*.
- Dopson, S., Ferlie, E., McGivern, G., Fischer, M. D., Mitra, M., Ledger, J. and Behrens, S. (2018) 'Leadership Development in Higher Education: A Literature Review and Implications for Programme Redesign', *Higher Education Quarterly*, 73(2), pp. 218–234.
- Du-Plessis, M. A. (2019) a Succession Planning Framework for Leadership Development of Nurse Educators in Nursing Education Institutions in South Africa.
- Ebbers, L., Conover, K. S. and Samuels, A. (2010) 'Leading from the Middle: Preparing Leaders for New Roles', *New Directions for Community Colleges*, (149), pp. 59–64.
- Eisenhardt, K. M. (1989) 'Building Theories from Case Study Research.', *Academy of Management Review*, 14(4), pp. 532–550.
- El-Alayli, A., Hansen-Brown, A. A. and Ceynar, M. (2018) 'Dancing Backwards in High Heels: Female Professors Experience More Work Demands and Special Favor Requests, Particularly from Academically Entitled Students', *Sex Roles*. Sex Roles, 79(3–4), pp. 136–150.

- Elamin, A. M. and Omair, K. (2010) 'Males' Attitudes Towards Working Females in Saudi Arabia', *Personnel Review*, 39(6), pp. 746–766.
- Fallatah, R. and Syed, J. (2017) Employee Motivation in Saudi Arabia: An Investigation into the Higher Education Sector, Springer.
- Fulton-calkins, P. and Milling, C. (2010) 'Community College Journal of COMMUNITY-COLLEGE LEADERSHIP: AN ART TO BE PRACTICED: 2010 AND BEYOND', (February 2015), pp. 37–41.
- Fusarelli, B. C., Fusarelli, L. D. and Riddick, F. (2018) 'Planning for the Future: Leadership Development and Succession Planning in Education', *Journal of Research on Leadership Education*, 13(3), pp. 286–313.
- Gaffney, S. (2005) 'Career Development as a Retention and Succession Planning Tool.', *The Journal for Quality and Participation*, 28(3), p. 7.
- Gandhi, M. and Sen, K. (2021) 'Missing women in Indian university leadership:

 Barriers and facilitators', *Educational Management Administration and Leadership*, 49(2), pp. 352–369.
- Gangone, L. and Lennon, T. (2014) 'Benchmarking women's leadership in academia and beyond', in *Women and leadership in higher education*, pp. 3–22.
- Garland, B. (2001) 'The Role of the Universities and CANTEACH in Succession Planning'.
- Gilbert, S. A. (2017) Succession Planning Relating to The Millennial Generation in Private Four-Year Universities.
- Glaser, B. G. (1978) Advances in the methodology of grounded theory: Theoretical sensitivity, University of California.
- Glaser, B. G. (1992) Basics of grounded theory analysis. Mill Valley.
- Glaser, B. G. (1998) Doing grounded theory: Issues and discussions. Sociology Press.
- Glaser, B. G. and Strauss, A. L. (1967) *Grounded theory: Strategies for qualitative research*. Chicago, lL: Aldine Publishing Company.
- Gola, C. H. and Bennett, M. H. (2016) 'The University of Houston's Liaison Services Advisory Board: A case study in leadership development and succession planning', College and Research Libraries News, 77(10), pp. 496– 501.
- Golafshani, N. (2003) 'Understanding Reliability and Validity in Qualitative Research', *The Qualitative Report*, 8(4), pp. 597–606.

- Goldman, C. (2020) 'Structure, Culture, and Agency: Examining Succession Planning in California State University (CSU) Libraries', *Journal of Library Administration*. Routledge, 60(1), pp. 1–21.
- Goulding, C. (1998) 'Grounded Theory: The Missing Methodology on the Interpretivist Agenda', *Qualitative Market Research: An International Journal*, 1(1), pp. 50–57.
- Grigoli, F., Koczan, Z. and Topalova, P. (2018) A Cohort-Based Analysis of Labor Force Participation for Advanced Economies.
- Guarino, C. M. and Borden, V. M. H. (2017) 'Faculty Service Loads and Gender: Are Women Taking Care of the Academic Family?', *Research in Higher Education*, 58(6), pp. 672–694.
- Hall-Ellis, S. D. and Grealy, D. S. (2013) 'The Dreyfus model of skill acquisition: A career development framework for succession planning and management in academic libraries', *College and Research Libraries*, 74(6), pp. 587–603.
- Hannon, P. (2014) Succession Planning: Developing The Next Generation of Nurse Leaders, Texas Woman's University.
- Harford, J. (2018) 'The perspectives of women professors on the professoriate: A missing piece in the narrative on gender equality in the university', *Education Sciences*, 8(2), pp. 1–17.
- Hartley III, H. V. and Godin, E. E. (2009) 'A Study of Career Patterns of the Presidents of Independent Colleges and Universities.', *Council of Independent Colleges*, p. 36.
- Heath, H. and Cowley, S. (2004) 'Developing a grounded theory approach: a comparison of Glaser and Strauss', 41, pp. 141–150.
- Hinojosa, J. L. (2019) 'The Athenaeum A Grounded Theory Study of Senior Leader Experiences and Perceptions of Unplanned Turnover'.
- Hodges, j. (2017) 'Cracking the walls of leadership: women in Saudi Arabia', Gender in management: an international journal, 32(1), pp. 34–46.
- Holton, J. A. (2010) 'The Coding Process and Its Challenges Judith', *The Grounded Theory Review*, 9(1), pp. 21–40.
- Howe-Walsh, L. and Turnbull, S. (2016) 'Barriers to women leaders in academia: tales from science and technology', *Studies in Higher Education*, 41(3), pp. 415–428.

- Hunt, E. A. (2009) 'Minding the Mission: Succession Planning for Academic Leadership at Christian Institutions'.
- Hunt, J. G. (Jerry) and Ropo, A. (1995) 'Multi-level Leadership: Grounded Theory and Mainstream Theory Applied to the Case of General Motors', *Leadership Quarterly*, 6(3), pp. 379–412.
- Hutchings, K., Lirio, P. and Metcalfe, B. D. (2012) 'Gender, globalisation and development: A re-evaluation of the nature of women's global work', *International Journal of Human Resource Management*, 23(9), pp. 1763– 1787.
- Ishak, A. K. and Kamil, B. A. M. (2016) 'Succession planning at higher education institutions: Leadership style, career development and knowledge management practices as its predictors', *International Review of Management and Marketing*, 6(7Special Issue), pp. 214–220.
- Jackson, N. C. and Dunn-Jensen, L. M. (2021) 'Leadership succession planning for today's digital transformation economy: Key factors to build for competency and innovation', *Business Horizons*. Elsevier Ltd, 64(2), pp. 273–284.
- Jacobs, J. A. and Winslow, S. E. (2004) 'The academic life course, time pressures and gender inequality', *Community, Work and Family*, 7(2), pp. 143–161.
- Jamjoom, F. B. and Kelly, P. (2013) 'Higher Education for Women in the Kingdom of Saudi Arabia Women 's Education in Saudi Society: A Brief Overview', pp. 117–125.
- Jones, P. and Rivers, R. (2021) 'Development of a Staff Nurse to Chief Nursing Officer Succession Planning Program', *Nurse Leader*. Elsevier Inc., 19(6), pp. 646–654.
- Kairys, M. R. (2018) 'The influence of gender on leadership in education management', *International Journal of Educational Management*, 32(5), pp. 931–941.
- Kallio, K. M., Kallio, T. J., Tienari, J. and Hyvönen, T. (2016) 'Ethos at stake: Performance management and academic work in universities', *Human Relations*, 69(3), pp. 685–709.
- Karam, C. M. and Afiouni, F. (2014) 'Localizing women's experiences in academia: multilevel factors at play in the Arab Middle East and North Africa', *International Journal of Human Resource Management*, 25(4), pp. 500–538.
- Kattan, M. M. (2015) Factors of Successful Women Leadership in Saudi Arabia.

- Keddy, B., Sims, S. L. and Stern, P. N. (1996) 'Grounded Theory as Feminist Research Methodology', *Journal of Advanced Nursing*, 23(3), pp. 448–453.
- Keller, K. (2018) 'Building the Case for Succession Planning in Higher Education: A Study of Succession Planning Pilots within the Minnesota State Colleges and Universities ...'.
- Kemp, L. J., Madsen, S. R. and El- Saidi, M. (2013) 'The current state of female leadership in the United Arab Emirates', *Journal of Global Responsibility*, 4(1), pp. 99–112.
- Kezar, A. J. (2017) 'Leadership in Higher Education, Concepts and Theories', Encyclopedia of International Higher Education Systems and Institutions.
- King, S. R. (2019) Succession Planning and Leadership Development In Texas Public Universities.
- Kirkland, T. P. (1990) A grounded theory of community college chief executive officer succession.
- Kitchenham, B., Pearl Brereton, O., Budgen, D., Turner, M., Bailey, J. and Linkman, S. (2009) 'Systematic Literature Reviews in Software Engineering A Systematic Literature Review', *Information and Software Technology*. Elsevier B.V., 51(1), pp. 7–15.
- Klein, M. F. and Salk, R. J. (2013) 'Presidential succession planning: A qualitative study in private higher education', *Journal of Leadership and Organizational Studies*, 20(3), pp. 335–345.
- Kocolowski, M. (2010) 'Shared leadership: Is it time for a change', *Emerging Leadership Journeys*, 3(1), pp. 22–32.
- Komives, S. R., Owen, J. E., Longerbeam, S. D., Mainella, F. C. and Osteen, L. (2005) 'Developing a leadership identity: A grounded theory', *Journal of College Student Development*, 46(6), pp. 593–611.
- Land, P. C. (2003) 'From the Other Side of the Academy to Academic Leadership Roles: Crossing the Great Divide', (124), pp. 13–20.
- Lapovsky, L. (2007) 'The Best-Laid Succession Plans.', Association of Governing Boards of Universities and Colleges, pp. 14–1.
- Larkin, J. and Neumann, R. (2012) 'Ageing Academics: Workforce Priorities for Universities', 20(1), pp. 3–24.
- Lerner, G. (1986) The creation of patriarchy (Vol.1), Women and History.

- Lin, M. (2019) 'Educational Thoughts of Western Radical Feminism', 15(3), pp. 21–25.
- Longman, K. A. and Anderson, P. S. (2016) 'Women in Leadership: The Future of Christian Higher Education', *Christian Higher Education*. Taylor & Francis, 15(1–2), pp. 24–37.
- Loomes, S., Owens, A. and McCarthy, G. (2019) 'Patterns of recruitment of academic leaders to Australian universities and implications for the future of higher education', *Journal of Higher Education Policy and Management*, 41(2), pp. 137–152.
- Lu, J., Laux, C. and Antony, J. (2017) 'Lean Six Sigma leadership in higher education institutions', *International Journal of Productivity and Performance Management*, 66(5), pp. 638–650.
- Luna, G. (2010) 'Succession planning: A doctoral program partnership for emerging community college leaders', *Community College Journal of Research and Practice*, 34(12), pp. 977–990.
- Luzebetak, A. K. (2010) 'Community college succession planning: Preparing the next generation of women for leadership roles.', *Dissertation Abstracts International Section A: Humanities and Social Sciences*, 71, p. 1885.
- Macfarlane, B. and Burg, D. (2018) 'Women Professors as Intellectual Leaders', in *Project report, Leadership Foundation for Higher Education*, pp. 1–40.
- Magner, D. K. (2009) 'The Leadership Pipeline', Chronicle of Higher Education.
- Mapolisa, T., Mhlanga, E., Chabvonga, N. and Chimbwadzwa, Z. (2013) 'Female Leadership Dilemmas in Primary Schools: A Case Study of Primary Schools in Harare Province in Zimbabwe', *Zimbabwe Journal of Educational Research*, 25(2), pp. 261–276.
- Martin, V. B. and Gynnild, A. (2011) *Grounded theory: The philosophy, method, and work of Barney Glaser*, 2011. Universal-Publishers.
- Mateso, P. E. . E. (2010) 'Understanding succession planning and management efforts at Midwestern University: a mixed methods study', p. 270.
- Mauthner, N. S. and Edwards, R. (2010) 'Feminist research management in higher education in Britain: Possibilities and practices', *Gender, Work and Organization*, 17(5), pp. 481–502.

- McMurray, A. M., Henly, D., Chaboyer, W., Clapton, J., Lizzio, A. and Teml, M. (2012) 'Leadership succession management in a university health faculty', *Journal of Higher Education Policy and Management*, 34(4), pp. 365–376.
- Meek, V. (2007) 'Succession planning and Australian higher education', in *The Treasure Within Leadership and Succession Planning*, pp. 111–125.
- Melton, L. C. (2018) 'Succession planning research in the educational sector', Succession Planning: Promoting Organizational Sustainability, pp. 175–186.
- Meriam, sharan b (2009) Qualitative Research A Guide to Design and Implementation Revised.
- Metcalfe, A. and Slaughter, S. (2008) 'The differential effects of academic capitalism on women in the academy', in *Unfinished agendas: New and continuing gender challenges in higher education*, pp. 80–111.
- Mifsud, D. (2019a) 'Introduction: The Positioning and Making of Female Academics—A Review of the Literature', in *The Positioning and Making of Female Professors: Pushing Career Advancement Open*, pp. 1–26.
- Mifsud, D. (2019b) 'Introduction: The Positioning and Making of Female Academics—A Review of the Literature', *Springer*.
- Miles, M. B., Huberman, M. a and Saldana, J. (2014) 'Drawing and Verying Conclusions', *Qualitative Data Analysis: A Methods Sourcebook*, pp. 275–322.
- Miller, J. (2010) 'Commentary on heidensohn's "the deviance of women": Continuity and change over four decades of research on gender, crime and social control', *British Journal of Sociology*, 61(SUPPL. 1), pp. 133–139.
- Miller, J. and Mullins, C. W. (2006) 'The Status of Feminist Theories in Criminology', in *Taking Stock*. Routledge, pp. 217–249.
- Míltersteíner, R. K., de Oliveira, F. B., Hryniewicz, L. G. C., Santanna, A. de S. and Moura, L. C. (2020) 'Female leadership: perceptions, reflections, and challenges in public administration', *Cadernos EBAPE*, 18(2), pp. 406–423.
- Mohnot, H. (2017) 'The impact of some demographic variables on academic leadership preparedness in Indian higher education', *Journal of Further and Higher Education*. Routledge, 9486(August), pp. 1–11.
- Morley, L. (2014) 'Lost leaders: Women in the global academy', *Higher Education Research and Development*, 33(1), pp. 114–128.

- Morrin, A. (2013) 'Factors that are important to succession planning: A case study of one Ontario college of applied arts and technology', *College Quarterly*, 16(1), pp. 1–18.
- Morse, J. M., Bowers, B. J., Charmaz, K., Corbin, J., Clarke, A. E. and Stern, P. N. (2016) *Developing grounded theory: The second generation, Routledge*.
- Moser, K. (2008) 'California's Community Colleges Struggle to Recruit and Retain President', *Chronicle of Higher Education*, 55(13).
- Moss-Racusin, C. A., Dovidio, J. F., Brescoll, V. L., Graham, M. J. and Handelsman, J. (2012) 'Science faculty's subtle gender biases favor male students', *Proceedings of the National Academy of Sciences of the United States of America*, 109(41), pp. 16474–16479.
- Mumford, M. D. (2000) 'Managing Creative People: Strategies and Tactics for Innovation', *Human Resource Management Review*, 10(3), pp. 313–351.
- Murray, R. and Mifsud, D. (2019) *The positioning and making of female professors:*Pushing career advancement open, Palgrave macmilla.
- Mustafa, R. F. and Troudi, S. (2019) 'Saudi Arabia and Saudi Women in Research Literature: A Critical Look', *Asian Social Science*, 15(2), p. 133.
- Ndunguru, G. (2018) 'The practice of Succession Planning in Tanzania's Higher Learning Institutions a Case of Local Government Training Institute (LGTI)', p. 115.
- Neefe, D. O. (2009) 'Succession Planning in a Two-Year Technical College System', A Dissertation Submitted to the Faculty of the Graduate School of the University of Minnesota, (May).
- Negrea, S. (2008) 'Succession planning: A necessity for community colleges', Institute for Community College Development.
- Nielsen, T. H. (1987) 'The Creation of Patriarchy', WOMEN AND HISTORY-LERNER, 1, pp. 292–232.
- Ollilainen, M. and Solomon, C. R. (2014) 'Carving a "Third Path": Faculty Parents' Resistance to the Ideal Academic Worker Norm', *Advances in Gender Research*, 19, pp. 21–39.
- Olson-Strom, S. and Rao, N. (2020) 'Higher Education for Women in Asia', in *Diversity and Inclusion in Higher Education: Lessons from accross Asia*, pp. 263–282.

- Omar, R. and Hassan, Z. A. (2015) 'Blooming Leaders Among Equals: a Case of Leadership Succession Planning in Malaysian Higher Education', Scientific Cooperations International Journal of Finance Business, Economics, Marketting and Information Systems, 1(1), pp. 30–40.
- Owolabi, T. and Adeosun, O. (2021) 'Succession Planning and Talent Retention: Evidence From the Manufacturing Sector in Nigeria', *British Journal of Management*, 4(January), pp. 17–32.
- Parkman, A. and Beard, R. (2008) 'Succession planning and the imposter phenomenon in higher education', *CUPA-HR Journal*, 59(2), pp. 29–36.
- Parry, K. W. (1998) 'Grounded Theory and Social Process: A New Direction for Leadership Research', *Leadership Quarterly*, 9(1), pp. 85–105.
- Pate, S. S. and Adams, M. (2013) 'The Influence of Social Networking Sites on Buying Behaviors of Millennials Part of the Advertising and Promotion Management Commons, Fashion Business Commons, Marketing Commons, and the Sales and Merchandising Commons', *Atlantic Marketing Journal*, 2(1), pp. 4–27.
- Patton, M. Q. (2002) *Qualitative research & evaluation methods (3rd ed.)*. Thousand Sage: Oaks.
- Petticrew, M. and Roberts, H. (2008) Systematic Reviews in the Social Sciences: A Practical Guide.
- Phillips, L. K. (2021) 'Succession planning in nursing education', *Nursing Outlook*. Elsevier Inc., 69(1), pp. 32–42.
- Pierce, D. (2015) 'Guiding light. How community college presidents can ensure their vision lives on, even after they move on', *Community College Journal*, 85(6), pp. 26–32.
- Power, A. (2020) 'Understanding leadership in higher education as a tool for change in relation to gender', in *Gender in academic recruitment and selection*, pp. 140–153.
- Probert, B. (2005) "I Just Couldn't Fit It In": Gender and Unequal Outcomes in Academic Careers', *Gender, Work and Organization*, 12(1), pp. 50–72.
- Raby, R. L. and Valeau, E. J. (2019) 'Position Training and Succession Planning for Community College International Education Leaders', *Community College Journal of Research and Practice*. Routledge, 45(2), pp. 86–102.

- Rich, D. (2006) 'Academic leadership and the restructuring of higher education', *New Directions for Higher Education*, 2006(134), pp. 37–48.
- Richards, C. L. (2009) A New Paradigm: Strategies For Succession Planning In Higher Education, Capella University.
- Richardsen, A. M., Traavik, L. E. M. and Burke, R. J. (2016) 'Women and work stress: more and different?', in *Handbook on Well-Being of Working Women*, pp. 123–140.
- Ritchie, M. (2020) 'Succession planning for successful leadership: Why we need to talk about succession planning!', *Management in Education*, 34(1), pp. 33–37.
- Röbken, H. (2007) 'Leadership Turnover Among University Presidents', *Management Revue*, 18(2), pp. 138–152.
- Rosenthal, J., Routch, K., Monahan, K. and Doherty, M. (2018) 'The holy grail of effective leadership succession planning: How to overcome the succession planning paradox', *Deloitte Insights*.
- Rothwell, W. (2010) Effective succession planning: Ensuring leadership continuity and building talent from within (4th ed.), Amacom. Amacom.
- Salari, S. and Abedini, M. (2017) 'Examining the role of management succession planning (MSP) in talent identification and empowerment of human capital in Hormozgan University of Medical Sciences', *Hormozgan Medical Journal*, 21(2), pp. 119–128.
- Saldana, J. (2011) Fundamentals of Qualitative Research.
- Santovec, M. Lou (2010) 'Succession Planning Gives Women a Shot at Leadership', *Women in Higher Education*. John Wiley & Sons, Ltd, 19(5), pp. 6–7.
- Saunders, B. and Kitzinger, J. (2015) 'Anonymising interview data: challenges and compromise in practice'.
- Savigny, H. (2014) 'Women, Know Your Limits: Cultural Sexism in Academia', in *Gender and Education*, pp. 794–809.
- Schueller-Weidekamm, C. and Kautzky-Willer, A. (2012) 'Challenges of work-life balance for women physicians/mothers working in leadership positions', *Gender Medicine*. Excerpta Medica, Inc., 9(4), pp. 244–250.
- Scott, H. M. P. (2007) The Temporal Integration of Connected Study into a Structured Life: A Grounded Theory.

- Semela, T., Bekele, H. and Abraham, R. (2020) 'Navigating the river Nile: the chronicle of female academics in Ethiopian higher education', *Gender and Education*. Taylor & Francis, 32(3), pp. 328–346.
- Shamsuddin, A., Chan, C., Sari, A. and Kassim, M. (2012) 'Leadership Management as an Integral Part of Succession Planning in HEIs: A Malaysian Perspective', *International Journal of Business and Social Science*, 3(3), pp. 151–159.
- Shava, G. N. and Tlou, F. N. (2018) 'Distributed Leadership in Education , Contemporary Issues in Educational Leadership', *African Educational Research Journal*, 6(4), pp. 279–287.
- Shaw, S. and Hernandez-gantes, V. (2021) 'The Status of Female Executive Leadership in Postsecondary Education', 32, pp. 1–16.
- Sherman, R., Dyess, S., Hannah, E. and Prestia, A. (2013) 'Succession planning for the future through an academic-practice partnership: A nursing administration master's program for emerging nurse leaders', *Nursing Administration Quarterly*, 37(1), pp. 18–27.
- Shults, C. (2001) 'The Critical Impact of Impending Retirements on Community College Leadership.', American Association of Community Colleges, Washington, DC.
- Sidani, Y. and Al Ariss, A. (2014) 'Institutional and Corporate Drivers of Global Talent Management: Evidence from the Arab Gulf Region', *Journal of World Business jo*, 49, pp. 215–224.
- Sidani, Y. M., Konrad, A. and Karam, C. M. (2015) 'From female leadership advantage to female leadership deficit A developing country perspective', *Career Development International*, 20(3), pp. 273–292.
- Simmons, J. A. (2009) "Both sides now": Aligning external and internal branding for a socially responsible era', *Marketing Intelligence and Planning*, 27(5), pp. 681–697.
- Song, W. and Hartley III, H. V. (2012) 'A study of presidents of independent colleges and universities', pp. 1–48.
- Spanier, G. B. (2010) 'The president as mentor.', AGB Trusteeship, 18(3).
- Stern, P. N. (2009) 'In the beginning Glaser and Strauss created grounded theory', in *Developing grounded theory: The second generation*, pp. 23–29.

- Stevens-Kalceff, M., Hagon, S., Cunningham, M. and Woo, A. (2007) *Maximising Potential in Physics: Investigation of the Academic Profile of the School of Physics at the University of New South Wales*.
- Straus, A. and Corbin, J. (1994) 'Grounded theory methodology: An overview', in *Handbook of qualitative research*, pp. 273–285.
- Stripling, J. and Medina, B. (2011) 'The graying presidency.', *The Chronicle of Higher Education*, 58(6), pp. 1–10.
- Swanson, M. A. (2018) 'Employee Perceptions of Succession Planning Within Higher Education a Qualitative Case Study', (May).
- Syed, J., Ali, F. and Hennekam, S. (2018) 'Gender equality in employment in Saudi Arabia: a relational perspective', *Career Development International*, 23(2), pp. 163–177.
- Tichy, N. M. (2014) Succession: Mastering the make or break process of leadership transition, go.gale.com. Penguin.
- Till, A., McKimm, J. and Swanwick, T. (2020) 'The importance of leadership development in medical curricula: A UK perspective (stars are aligning)', *Journal of Healthcare Leadership*, 12, pp. 19–25.
- Tlaiss, H. and Kauser, S. (2011) 'The Impact of Gender, Family, and Work on the Career Advancement of Lebanese Women Managers', *Gender in Management: An International Journal*, 26(1), pp. 8–36.
- Torabi, N., Karimi, F. and Nadi, M. A. (2019) 'Strategies For Succession Planning In Universities: A Mixed-Methods Analysis', *MALAYSIAN ONLINE JOURNAL OF EDUCATIONAL MANAGEMENT (MOJEM)*, 7(2), pp. 46–71.
- Tran, T. Q. (2020) 'Identifying female leadership and performance in small and medium-sized enterprises in a transition economy: The case study of Vietnam', *Asian Economic and Financial Review*, 10(2), pp. 132–145.
- Tranfield, D., Denyer, D. and Smart, P. (2003) 'Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review', *British Journal of Management*, 14, pp. 207–222.
- Turner, P. (2018) 'The Role of Leaders: The Importance of Leadership', *Leadership in Healthcare*, (Yukl 2010), pp. 45–74.
- UNDP (2020) '2020 HDR Technical Notes', pp. 1–19.

- Varshney, D. (2019) 'The Strides of the Saudi female workforce: Overcoming constraints and contradictions in transition', *Journal of International Women's Studies*, 20(2), pp. 359–372.
- Walker, D. and Myrick, F. (2006) 'Qualitative Health Research', *Qualitative health* research, 16(4), pp. 547–559.
- Washington, C. E. (2016) 'Succession planning in higher education'.
- De Welde, K., Ollilainen, M. and Solomon, C. R. (2019) 'Feminist Leadership in the Academy: Exploring Everyday Praxis', *Advances in Gender Research*, 27, pp. 3–21.
- De Welde, K. and Stepnick, A. (2015) 'From People to Policies: Enduring Inequalities and Inequities for Women Academics', in *Disrupting the culture of silence: Confronting gendDe Welde, K. and Stepnick, A. (2015) 'From People to Policies: Enduring Inequalities and Inequities for Women Academics', in Disrupting the culture of silence: Confronting gender inequality and making cha, pp. 1–28.*
- Wilkins, S. and Neri, S. (2019) 'Managing Faculty in Transnational Higher Education: Expatriate Academics at International Branch Campuses', *Journal of Studies in International Education*, 23(4), pp. 451–472.
- World Economic Forum (2018) 2018 The Global Gender Gap Report, World Economic Forum.
- Wright, R. T., Campbell, D. E., Thatcher, J. B., Roberts, N., Wright, R. T. and Roberts, N. (2012) 'Operationalizing Multidimensional Constructs in Structural Equation Modeling: Recommendations for IS Research Modeling: Recommendations for IS Research', 30.
- Wu, Q., Cormican, K. and Chen, G. (2020) 'A Meta-Analysis of Shared Leadership: Antecedents, Consequences, and Moderators', *Journal of Leadership and Organizational Studies*, 27(1), pp. 49–64.
- Yin, R. K. (2003) 'Designing case studies', *Qualitative research methods*, 5(14), pp. 359–386.

Appendix A Instrument and interview protocol

Interview Protocol for Informants in Universities (Insider)

Research	Interview Questions	Time
questions		
In universities, key informants are: board of trustees, senior management person		
(known as rector, director, deputy-vice chancellor, chancellor), dean, head of		
departments (known as deputy/vice dean), instructors (known as faculty/lecturers),		
administrator ((line managers or operational staff)	
Introduction	Interviewee's basic information:	2 mins
	1) Name:	
	2) University:	
	3) Position:	
	4) Qualification:	
	5) Experience:	
	6) Nationality:	
	7)Gender:	
	8)Contact number/email:	
RQ1: How	Style of leadership being practiced?	15
does	Specification: to know and understand the current style of	minute
leadership is	leadership in higher education	S
perceived in		
Saudi	• What does leadership look like in your	
Arabian	university?	
universities?	Probe: How do you define leadership?	
	• How do leaders in your university manage	
	operations and people?	
	Probe: look at how the process of leadership is being practiced	

RQ2: How	Female leadership in terms of what, how, and why?	45
is the current situation of female leadership in Saudi	Specification: to know and understand the perception of people in higher education; to know the importance of female leaders in higher education and influence to strengthen the educational systems	mins
Arabian	How do you see the female role as a leader in	
universities?	your university?	
	Probe: Aside from Characteristics of female leaders, we want to know how the university sees the female staff are able to accentuate/demonstrate their leadership through their duties/responsibilities/positions/personalities (and other aspects related to roles)-	
	• Do you think that your university should have	
	female leaders? And why?	
	Probe: Assumptions and reasons for the need for female leaders in universities	
	 Practices and cultural aspects that support or 	
	suppress female leadership?	
	Specification: to explore current practices in terms of female leadership in higher education within faculties or schools, diversity, and inclusion, opportunity to participate and lead programs/events/conferences,	
	• What are the current practices that support	
	female leadership in your university?	
	Probe: Do females work at leadership positions in your university? If yes, then what kinds of tasks are they taking care of?	
	• What are the cultural aspects that support	

female leadership in your university?

Probe: Do females work at leadership positions in your university? If yes, then what kinds of tasks are they taking care of?

• What are the current practices that suppress female leadership in your university?

Probe: Do females face a tough time rendering services at leadership positions in your university? If yes, then what are those practices that suppress females to take aside from leadership roles?

• What are the cultural aspects that suppress female leadership in your university?

Probe: Do females face a tough time rendering services at leadership positions in your university? If yes, then what are those practices that suppress females to take aside from leadership roles?

Acceptance of female leaders?

Specification: the environment in higher education, work dynamics, social, cultural, and religious aspects, capabilities to handle the leadership role in higher education

• Do people accept female leaders at your university?

Probe: How and why?

 How do male workers perceive the female role as a leader in your university?

Probe: If perceived positively, then ask why?

If perceived negatively, then ask why?

RQ3: To Succession planning in terms of what, how, and why?

what extend does succession planning is contributing to the advancemen t of female leaders in Saudi Arabian universities?

Specification: to know the level of understanding of people about succession planning; in business organizations and higher education; Definition and practical know-how about succession planning

30 min

 Does your university have a specific process for identifying and developing new leaders who can replace old leaders when they leave, retire or die?

Probe: Does your university have succession planning? How will you define succession planning, and what are important things that should be part of succession planning in your university?

 How are future leaders or top management positions are chosen to lead in your university?

Probe: Components of succession planning in the university.

Female Succession Planning in terms of what, how, and why?

Specification: to know the level of understanding of people about female succession planning; in business organizations and higher education; Definition and practical know-how about female succession planning

Does/How female staff is included in the process of choosing the next dean/vice chancellor/president?

Probe: We would like to see how female staff are included either as a panel for the succession planning/candidates/mentors/

coaches in the process of developing future bosses/top management of the universities.

Development of female succession planning?

Specification: to know acceptance level to execute female succession planning, key steps involved in the development of female succession planning in higher education; to know challenges and overcome in execution of female succession planning in higher education

• What are the challenges to nurturing female leaders in your institutions?

Probe what are the hurdles that they face to adopt succession planning in your university?

• How can the university develop a process for female staff to be fully considered in the succession planning in your university?

Probe: Organize a workshop to design university-specific design for female succession planning, deal with people perceptions via promoting a culture of female leaders, a consensus of senior management to engage female talents in a leadership role as a part of succession planning

Total duration 92 mins

Interview Protocol for Informants Outside Universities (outsider)

Research	Interview Questions	Time	
questions			
Outside universities, key informants are: 1. education management consultants (who			
provide consultancies to universities and institutions for development and connect			
with education	ministries. 2. Senior personnel from KSA education ministry	(who	
monitor the operations in universities)			
Introduction	Interviewee's basic information:	2 mins	
	1. Name:		
	2. University:		
	3. Position:		
	4. Qualification:		
	5. Experience:		
	6. Nationality:		
	7. Gender:		
	8. Contact number/email:		
RQ1: How	Style of leadership being practiced?		
does	Specification: to know and understand the current style of		
leadership is	leadership in higher education		
perceived in			
Saudi	• What does leadership look like in the		
Arabian	universities?		
universities?	Probe: How do you define leadership?		
	• How do leaders in the universities manage		
	operations and people?		
	Probe: look at how the process of leadership		
	is being practiced		

RQ2: How is	Female leadership in terms of what, how, and why?	20
the current	Specification: to know and understand the perception of	mins
situation of	people in higher education; to know the importance of	
female	female leaders in higher education and influence to	
leadership in	strengthen the educational systems	
Saudi	,	
Arabian	How do you see the female role as a leader in the	
universities?	universities?	
	Probe: Aside from Characteristics of female	
	leaders, we want to know how the university	
	sees the female staff are able to	
	accentuate/demonstrate their leadership	
	through their	
	duties/responsibilities/positions/personalities	
	(and other aspects related to roles)-	
	Do you think that universities should have female	
	leaders? And why?	
	Probe: Assumptions and reasons for the	
	need for female leaders in universities	
	Practices and cultural aspects that support or suppress	25
	female leadership?	mins
	Specification: to explore current practices in terms of	
	female leadership in higher education within faculties or	
	schools, diversity, and inclusion, opportunity to participate	
	and lead programs/events/conferences,	
	What are the current practices that support	
	female leadership in universities?	
	Probe: Do females work at leadership	
	positions in universities? If yes, then what	
	kinds of tasks are they taking care of?	

• What are the cultural aspects that support female leadership in the university?

Probe: Do females work at leadership positions in the university? If yes, then what kinds of tasks are they taking care of?

• What are the current practices that suppress female leadership in universities?

Probe: Do females face a tough time rendering services at leadership positions in universities? If yes, then what are those practices that suppress females to take aside from leadership roles?

• What are the cultural aspects that suppress female leadership in the university?

Probe: Do females face a tough time rendering services at leadership positions in the university? If yes, then what are those practices that suppress females to take aside from leadership roles?

Acceptance of female leaders?

Specification: the environment in higher education, work dynamics, social, cultural, and religious aspects, capabilities to handle the leadership role in higher education

• Do people accept female leaders in universities?

Probe: How and why?

• How do male workers perceive the female role as a

	leader in the universities?	
	Probe: If perceived positively, then ask why?	
	If perceived negatively, then ask why?	
RQ3: To	Succession planning in terms of what, how, and why?	30
what extend	Specification: to know the level of understanding of people	mins
does	about succession planning; in business organizations and	
succession	higher education; Definition and practical know-how about	
planning is	succession planning	
contributing		
to the	Does the university have a specific process for	
advancement	identifying and developing new leaders who can	
of female	replace old leaders when they leave, retire or	
leaders in	die?	
Saudi	Probe: Do universities have succession	
Arabian	planning? How will you define succession	
universities?	planning, and what are important things that	
	should be part of succession planning in the	
	universities?	
	How are future or top management positions are	
	chosen to lead in the universities?	
	Probe: Components of succession planning	
	in the universities.	
	Female Succession Planning in terms of what, how, and why?	
	Specification: to know the level of understanding of people	
	about female succession planning; in business	
	organizations and higher education; Definition and	
	practical know-how about female succession planning	
	Does/How female staff is included in the process	

of choosing the next dean/vice chancellor/president?

Probe: We would like to see how female staff are included either as a panel for the succession planning/candidates/mentors/coaches in the process of developing future bosses/top management of the universities.

Development of female succession planning?

Specification: to know acceptance level to execute female succession planning, key steps involved in the development of female succession planning in higher education; to know challenges and overcome in execution of female succession planning in higher education

• What are the challenges to nurturing female leaders in the universities?

Probe: What are the hurdles they face to adopting succession planning in the universities?

 How can the university develop a process for female staff to be fully considered in the succession planning in the universities?

Probe: Organize a workshop to design university-specific design for female succession planning, deal with people perceptions via promoting a culture of female leaders, a consensus of senior management to engage female talents in a leadership role as a part of succession planning

Total duration

92

mins

LIST OF PUBLICATIONS

Dahlan, Dina; Omar, Rosmini; Kamarudin, Suzilawati; Nasar, Asim (2021) A systematic review of succession planning in higher education. *International Journal of Advanced and Applied Sciences*, 8(12)

Dahlan, D.; Omar, R.; Kamaruddin, S.; and Mohammadi, P. (2021) Female Leadership in Saudi Arabia's Higher Education: Exploring the Challenges and Barriers. *Review of International Geographical Education (RIGEO)*, 11(12)